

City of Cayce Regular Council Meeting Tuesday, October 1, 2019 6:00 p.m. - Cayce City Hall - 1800 12th Street cavcesc.gov

1. Call to Order

- A. Invocation and Pledge of Allegiance
- B. Approval of Minutes September 3, 2019 Regular Meeting September 18, 2019 Regular Meeting
- II. Public Comment Regarding Items on the Agenda

III. Ordinances and Resolutions

- Discussion and Approval of Ordinance 2019-16 Amending the Sewer Use Α. Ordinance - Second Reading
- B. Discussion and Approval of Ordinance 2019-17 Amending City Code Section 40-92 ("Prohibited Discharges") of Article III ("Sewer System") of Chapter 40 ("Utilities") - Second Reading
- C. Discussion and Approval of Ordinance 2019-18 Amending Zoning Ordinance Article 2 Definitions, Article 8 Table 8-B Number, Dimension and Location of Signs Requiring a Permit and Article 8 Table 8-B Table Notes to Add Language Concerning Digital Display Signs - First Reading

IV. Other

- A. Discussion and Approval of Memorandum of Understanding and Agreement between City of Cayce and Town of Springdale for Building Official Services
- Discussion and Approval of Bid Award for Construction of Art Lot B.
- C. Discussion and Approval of a Proposal for Cayce Arts District Design and Management

V. City Manager's Report

Committee Matters VI.

A. Approval to enter the following Committee approved Minutes into the City's Record Zoning Board of Appeals - January 28, 2019

Planning Commission – August 19, 2019 Events Committee – August 22, 2019

VII. Council Comments

VIII. Executive Session

- A. Receipt of legal advice relating to claims and potential claims by and against the City and other matters covered by the attorney-client privilege
- B. Discussion of possible contract negotiations for economic development concerning the City of Cayce
- IX. Reconvene
- X. Possible Actions by Council in follow up to Executive Session
- XI. Adjourn

SPECIAL NOTE: Upon request, the City of Cayce will provide this document in whatever form necessary for the physically challenged or impaired.

Mayor

Mayor Pro-Tem Elise Partin Tara S. Almond Council Members Phil Carter Eva Corley James E. Jenkins City Manager Tracy Hegler Assistant City Manager James E. Crosland



CITY OF CAYCE Regular Council Meeting September 3, 2019

The September Regular Council Meeting was held this evening at 6:00 p.m. in Council Chambers. Those present included Mayor Elise Partin, Council Members Tara Almond, Phil Carter, Eva Corley and James Jenkins. City Manager Tracy Hegler, Assistant City Manager Jim Crosland, Municipal Clerk Mendy Corder, Municipal Treasurer Garry Huddle, Planning and Development Director Carroll Williamson, Utilities Director Blake Bridwell, Director of Public Safety Byron Snellgrove, Deputy Director of Public Safety JJ Jones, Human Resources Director Lynn Dooley and City Attorney Danny Crowe were also in attendance.

Mayor Partin asked if members of the press and the public were duly notified of the Council Meeting in accordance with the FOIA. Ms. Corder confirmed they were notified.

Call to Order

Mayor Partin called the meeting to order. Council Member Almond gave the invocation. Mayor Partin stated that there were some special guests in attendance. Mr. Ken Parsons stated that he and Cub Scouts Hayden Eargle, Campbell Parsons, Graham Parsons and Jennings Parsons were with Cub Scout Pack 331. The young men introduced themselves, joined Council on the dais and led the assembly in the Pledge of Allegiance.

Approval of Minutes

Council Member Jenkins made a motion to approve the August 6, 2019 Regular Council Meeting minutes and the August 21, 2019 Regular Council Meeting minutes as written. Council Member Carter seconded the motion which was unanimously approved by roll call vote.

Public Comment Regarding Items on the Agenda

No one signed up for Public Comment.

Presentations

A. Presentation by Mr. John Sharpe re debris on roads

Mr. John Sharpe stated that he has experienced numerous issues with trucks leaving Martin Marietta, driving down Frink Street and dropping rocks on the

roadway. He stated that his business is located on Frink Street and often trucks pull over in front of his business to adjust their rock load and the rocks fall into the street or in his parking lot. He stated that some of the truck drivers told him that Martin Marietta would not allow them to adjust their load at the rock quarry so they have to pull over once they leave and adjust their load.

Mr. Sharpe stated that he has spoken to the truck drivers, the companies they work for, Martin Marietta staff, Cayce Public Safety and the State Transport Police and the issue has not improved. He stated that the City needs an Ordinance that requires each truck to be properly loaded and the load secured before the truck can leave Martin Marietta. Mr. Sharpe stated that he thinks Cayce Public Safety should patrol Frink Street a couple of times a week and pull trucks over that have rocks falling from them. He stated that the Ordinance should allow Officers to fine the truck driver, the company the driver works for and Martin Marietta.

B. Presentation of Municipal Association of South Carolina Achievement Award to the City of Cayce for Waterline Replacement Project from the Municipal Association of South Carolina

Mr. Jeff Shacker with the Municipal Association of South Carolina stated he was in attendance to award Council a trophy for the City winning a MASC Achievement Award in the Public Works category for the City's Waterline Replacement Project.

Mr. Shacker played a video detailing the City's Waterline Replacement Project. It stated that the City faced growing challenges with its water system with issues like water discoloration, flow and pressure problems, as well as line breaks. Much of the City's infrastructure was built from the 1930s to the 1950s with galvanized, cast iron and asbestos cement piping. System replacement is often handled incrementally, but since the City qualified for a low-interest State Revolving Fund loan, staff created an ambitious \$29 million plan, the largest project in the City's history, to handle all the needed replacements at the same time to minimize the period of disruption. The City also enlisted the Cayce Utility Fund as well as a lease purchase agreement backed by that fund to finance the work. After two years of design work, construction began in April 2017 and made use of seven contractors. When the project was completed in December 2018, crews had replaced 262,000 linear feet of existing water lines, 4,000 water meters, fire hydrants and a 1-million-gallon elevated storage tank. The completed project represents 75 percent of the City's water infrastructure. Communication efforts formed a key part of the project, emphasizing that the City wanted transparency and valued business concerns and customer satisfaction. The City's engineer was contracted to handle all resident communications and complaints. Contracting out for a dedicated customer service phone, email and communication system allowed City staff to continue providing routine services. Councilmembers informed residents and gave

updates at neighborhood meetings and community events, and all City staff carried Waterline Replacement Project business cards.

Mr. Shacker presented Council with the award and stated that it was an amazing project.

C. Presentation of the 2019 South Carolina Municipal Elected Officials Institute of Government's Honor Roll to Mayor and Council from the Municipal Association of South Carolina

Mr. Jeff Shacker stated that the Municipal Association of South Carolina's Honor Roll recognizes annually the city councils that can count all members of their sitting council as graduates of the Municipal Elected Officials Institute. He stated that the City is one (1) of only 18 municipalities out of 271 in the state on the Honor Roll. He stated that to graduate from the MEO Institute officials participate in more than 25 hours of classroom work. Highlights of topics covered during the sessions include the Freedom of Information Act, ethics and accountability, budgeting, planning and zoning, human resources and economic development. Mr. Shacker congratulated Council on this accomplishment.

Ordinances and Resolutions

A. Discussion and Approval of Ordinance 2019-12 Amending Budget Ordinance Adopted June 19, 2019 (Ordinance 2019-10) to Correct Scrivener's Errors as to Millage – Second Reading

Council Member Corley made a motion to approve Ordinance 2019-12 on Second Reading. Council Member Almond seconded the motion which was unanimously approved by roll call vote.

B. Discussion and Approval of Ordinance 2019-13 Amending Zoning Ordinance Section 6.2 ("Purpose of Districts") and Section 6.10 ("Design Overlay District") to Provide for a New Arts Design Overlay District – First Reading

Ms. Hegler stated that Council approval was requested for the First Reading of an Ordinance to amend the Zoning Ordinance to create an Arts Design Overlay District. She stated it would permit additional uses and apply design standards that are favorable to artists and others who would like to live and work in the same location. The development standards are meant to further enhance a pedestrian-friendly downtown community.

Ms. Hegler stated that staff is proposing a change to the Zoning Ordinance to create a design overlay for the area on State Street from Poplar Street south to Frink Street and Frink Street from State Street west to Foreman Street, including those properties on the western side of the intersection of Foreman Street and Frink Street. The overlay district extends to the rear property line of all properties contiguous to the right-of-way of Frink Street and State Street within these boundaries.

Ms. Hegler stated that the Planning Commission considered the zoning request for creation of this Design Overlay District at their meeting on August 19, 2019. The zoning request was opened to the public. Approximately 30 people were in attendance with several members of the public speaking both for and against. The largest concern was the parking location requirement. The Planning Commission recommended and staff agreed to modify the language to only require parking on the side or rear yard when new structures are added to the property. Additionally, based on public comment, the Planning Commission recommended and staff agreed to modify the boundaries of the Overlay District to include the entirety of each contiguous lot, rather than a maximum number of feet from the right-of-way, as was originally proposed. One member of the public spoke against the no minimum parking requirement because of the concern over parking spillover onto his property from an adjacent property. The Planning Commission voted on the zoning request to create the Arts Design Overlay District at the Planning Commission meeting on August 19, 2019.

Council Member Almond made a motion to approve Ordinance 2019-13 on First Reading. Council Member Jenkins seconded the motion which was unanimously approved by roll call vote.

 C. Discussion and Approval of Ordinance 2019-14 Amending Zoning Map and Rezoning Property to Create a New Arts Design Overlay District – First Reading

Council Member Almond made a motion to approve Ordinance 2019-14 amending the zoning map and rezoning property to create a new Arts Design Overlay District. Council Member Corley seconded the motion which was unanimously approved by roll call vote.

D. Discussion and Approval of Ordinance 2019-15 Amending Zoning Ordinance Section 7.5 Temporary Uses to Add Language Concerning Halloween – related Events – First Reading

Ms. Hegler stated that recently several requests have been made to City staff concerning Halloween-related events like haunted houses and escape rooms. To permit this use for an extended period of time leading up to Halloween, the temporary use section

of the Zoning Ordinance needs more specific language. This text amendment would specifically permit Halloween-related events from late September until November 1 as a temporary use in all commercial and industrial zoning districts. She stated that in order to reduce the impact to the surrounding properties, the text amendment would also require that these events be held completely inside of an enclosed building and must have adequate parking on site to meet this type of use.

Ms. Hegler stated that as with all temporary uses, the permit would require the approval of the Planning Director, Public Safety, the Fire Marshal, and the Building Official. The Planning Commission met on August 19, 2019, to hear Public Comment about the suggested changes. No one from the public spoke on this text amendment. The Planning Commission voted on the requested text amendment at the planning commission meeting on August 19, 2019.

Council Member Corley made a motion to approve Ordinance 2019-15 on First Reading. Council Member Almond seconded the motion which was unanimously approved by roll call vote.

Other

A. Discussion and Motion to Ratify Action of City Manager

Ms. Hegler stated that Council's approval was needed in order to ratify an agreement executed by her, under the Emergency Powers of Sec. 2-146, utilizing emergency powers.

Council Member Jenkins made a motion to ratify the City Manager's action executing an agreement using Emergency Powers. Council Member Carter seconded the motion which was unanimously approved by roll call vote.

City Manager's Report

Ms. Hegler stated that Council and the assembly may have noticed the beautiful wraps on utility boxes around the City showcasing some of the City's iconic images. She stated this project was funded by the Cayce Beautification Foundation and Swatch Graphics did the work. She stated that there would also be wraps on the traffic signal cabinets on Knox Abbott Drive. She stated that the Art Lot Public Input Session was very successful. It was held the prior week at Piecewise Coffee to go over the plans for the Art Lot at 1908 State Street. Ms. Hegler stated that the concept is developing nicely and staff is pushing to have it completed in time for Engenuity's Competiveness Week at the end of January, 2020. She stated that Jaron King, a first year graduate student at the University of South Carolina will be starting as an intern with the City on Monday, September 9. Mr. King will be working with the City to develop a Parks, Recreation and

Cultural/Arts Plan. Ms. Hegler stated that staff was monitoring Hurricane Dorian closely. She stated that she watched the Governor's press conference that afternoon and they're still not sure of the storm's path, though it has weakened. She stated that the City remains prepared nonetheless.

Committee Matters

A. Approval to enter the following Committee approved Minutes into the City's Record

Events Committee – June 13, 2019

Cayce Housing Authority – June 18, 2019

Planning Commission – July 15, 2019

Council Member Almond made a motion to enter the approved Events Committee, Cayce Housing Authority and Planning Commission meeting minutes into the record. Council Member Jenkins seconded the motion which was unanimously approved by roll call vote.

Council Comments

Council Member Jenkins stated that he hoped Hurricane Dorian would not affect the state and that everyone would be safe.

Executive Session

- A. Receipt of legal advice relating to claims and potential claims by and against the City and other matters covered by the attorney-client privilege
- B. Discussion of possible contract negotiations for economic development concerning the City of Cayce

Mayor Partin stated that there was not anything to be discussed under Item IX. A. Council Member Jenkins made a motion to move into Executive Session to discuss Item IX. B. Council Member Carter seconded the motion which was unanimously approved by roll call vote.

Reconvene

After the Executive Session was concluded, Council Member Almond made a motion to reconvene the Regular meeting. Council Member Corley seconded the motion which was unanimously approved by roll call vote.

City of Cayce Minutes of 9/3/2019 Regular Council Meeting Page 7

Possible Actions by Council in follow up to Executive Session

No action was taken in follow up to Executive Session.

Adjourn

There being no further business, Council Member Almond made a motion to adjourn the meeting. Council Member Corley seconded the motion which was unanimously approved by roll call vote. The meeting adjourned at 7:15 p.m.

ATTEST:	Elise Partin, Mayor	
Mendy C. Corder, CMC, Municipal Clerk		

IF YOU WOULD LIKE TO SPEAK ON A MATTER APPEARING ON THE MEETING AGENDA, PLEASE COMPLETE THE INFORMATION BELOW PRIOR TO THE START OF THE MEETING.* THANK YOU.

COUNCIL MEETING SPEAKERS' LIST

Date of Meeting September 3, 2019

Name	Address	Agenda Item

^{*}Appearance of citizens at Council meetings - City of Cayce Code of Ordinances, Sec. 2-71. Any citizen of the municipality may speak at a regular meeting of the council on a matter pertaining to municipal services and operation, with the exception of personnel matters, by notifying the office of the city manager at least five working days prior to the meeting and stating the subject and purpose for speaking. Additionally, during the public comment period as specified on the agenda of a regular meeting of the council, a member of the public may speak on a matter appearing on the meeting agenda, with the exception of personnel matters by signing a speakers list maintained by the city clerk prior to the start of the public comment period. At the discretion of the mayor or presiding officer, the length of time for any speaker's presentation may be limited and the number speakers also may be limited.

James E. Jenkins



City of Cayce Regular Council Meeting September 18, 2019

A Regular Council Meeting was held this afternoon at 5:00 p.m. in Council Chambers. Those present included Mayor Elise Partin, Council Members Tara Almond, Phil Carter, Eva Corley and James Jenkins. City Manager Tracy Hegler, Assistant City Manager Jim Crosland, Municipal Clerk Mendy Corder, City Treasurer Garry Huddle, Planning and Development Director Carroll Williamson, Director of Public Safety Byron Snellgrove, Deputy Director of Public Safety JJ Jones, Director of Human Resources Lynn Dooley and City Attorney Danny Crowe were also in attendance.

Mayor Partin asked if members of the press and the public were duly notified of the meeting in accordance with the FOIA. Ms. Corder confirmed they were notified.

Call to Order

Mayor Partin called the meeting to order and Council Member Corley gave the invocation. Mayor Partin led the assembly in reciting the Pledge of Allegiance.

Public Comment Regarding Items on the Agenda

Ms. Corder stated no one had signed up for public comment.

Presentations

- A. Presentation by Mr. Robert Evans re the City's Waterline Replacement Project
- Mr. Evans voiced concern that Riverland Park was not included in the City's Waterline Replacement Project.

Ordinances and Resolutions

A. Discussion and Approval of Ordinance 2019-13 Amending Zoning Ordinance Section 6.2 ("Purpose of Districts") and Section 6.10 ("Design Overlay District") to Provide for a New Arts Design Overlay District – Second Reading

Council Member Almond made a motion to approve Ordinance 2019-13 on Second Reading. Council Member Corley seconded the motion which was unanimously approved by roll call vote.

B. Discussion and Approval of Ordinance 2019-14 Amending Zoning Map and Rezoning Property to Create a New Arts Design Overlay District – Second Reading Council Member Corley made a motion to approve Ordinance 2019-14 on Second Reading. Council Member Almond seconded the motion which was unanimously approved by roll call vote.

C. Discussion and Approval of Ordinance 2019-15 Amending Zoning Ordinance Section 7.5 Temporary Uses to Add Language Concerning Halloween-related Events – Second Reading

Council Member Carter made a motion to approve Ordinance 2019-15 on Second Reading. Council Member Jenkins seconded the motion which was unanimously approved by roll call vote.

D. Discussion and Approval of Ordinance 2019-16 Amending the Sewer Use Ordinance – First Reading

Ms. Hegler stated that Council approval was needed to revise the Sewer Use Ordinance (SUO) by removing the separate Fats, Oils, and Grease (FOG) Control Policy. She stated that approval for a revision to City Ordinance Section 40-92 was also needed for enforcement purposes. The existing SUO sets forth uniform requirements for users of the Publicly Owned Treatment Works for the City and enables the City to comply with all applicable State and Federal laws.

Ms. Hegler stated that the SUO was revised as of November 2018. The revised SUO includes the removal of the FOG Control Policy in the Appendix, however the FOG Control Policy is referenced several times in the revised SUO. She stated that the City Attorney and DHEC approved the revised SUO and DHEC recommends the segregation of the SUO and FOG Control Policy so that the City can adjust the FOG Control Policy without further governmental approvals, when conditions warrant. She stated that Governmental approvals are still required for revisions to the SUO.

Ms. Hegler stated that the FOG Control Policy is to aid staff in the administration of the City's Fats, Oils, and Grease Ordinance ("FOG Ordinance") designed to reduce the introduction and accumulation of fats, oils, and grease into the municipal wastewater collection system. This policy establishes uniform permitting, maintenance and monitoring requirements for controlling the discharge of grease from food service facilities discharging into the wastewater collection system and for regulation of commercial grease haulers operating within the City's collection system. Ms. Hegler stated that this policy was last revised in July 2016. It was a part of the aforementioned SUO, also adopted and approved by City Council at that time. She stated that the FOG Control Policy will be a separate document. As a policy, the FOG Control Policy can be changed and modified without further approvals by City Attorney, DHEC, or City Council.

Council Member Almond made a motion to approve Ordinance 2019-16 on First Reading. Council Member Corley seconded the motion which was unanimously approved by roll call vote.

E. Discussion and Approval of Ordinance 2019-17 Amending City Code Section 40-92 ("Prohibited Discharges") of Article III ("Sewer System") of Chapter 40 ("Utilities") – First Reading

Council Member Almond made a motion to approve Ordinance 2019-17 on First Reading. Council Member Corley seconded the motion which was unanimously approved by roll call vote.

F. Consideration and Approval of Resolution Supporting a Grant Application from the City of Cayce to the Municipal Association of South Carolina

Ms. Hegler stated that staff plans to submit a grant application to the Municipal Association of South Carolina (MASC) for the Hometown Economic Development Grant opportunity. As part of the grant application, the City is required to include a Resolution of Council in support of the application. She stated that MASC has opened a grant opportunity for cities and towns in South Carolina to implement economic development projects that will make a positive impact on the quality of life in their communities. Ms. Hegler stated that staff plans to apply for a grant and request \$25,000 in funding. Grant funds would be used for establishment of an arts and culture district. Establishment of an arts district will benefit both the City and stores and artists. She stated that the arts district will include allowing live/work uses throughout the district for properties, flexibility in design, public art pieces and distinct signage. She stated that parking requirements will be reduced and flexible parking options will also be available.

Ms. Hegler stated that the grant requires a 15% match by the City. The total project will be \$28,750 to implement, of which \$25,000 would be grant funds and \$3,750 would be City match. She stated that the grant application is currently in development and will be submitted by the due date of September 27, 2019.

Council Member Carter made a motion to approve the Resolution supporting the grant application. Council Member Almond seconded the motion which was unanimously approved by roll call vote.

City Manager's Report

Ms. Hegler stated that she was pleased to announce that a new Grants/Projects Manager was hired. She stated that Ms. Sarah Harris is thrilled to be starting with the City on October 3. Ms. Harris has experience in local, state and federal grant management and currently works for the SC Department of Public Safety. Ms. Hegler stated that a Call for Artists was issued for two murals on State Street (at 1804 and 2005). She stated that staff received an astounding 20 responses to the Call and were

currently going through them. Ms. Hegler stated that the timeline to select the artist/s was by the end of that week. She stated that recently some of the traffic signal cabinet's in the City were wrapped with photos of iconic images in the City.

Council Comments

Council Member Almond stated that the Design Overlay District item on the agenda caused her to think about other design overlay districts in the City. She asked staff to take a look at the other design overlay districts to ensure that the prohibited uses and permissible uses are still inline since there are new types of businesses since the overlay districts were created.

Council Member Carter asked Ms. Hegler to thank the Sanitation Department for him. He stated that a neighbor of his was moving and got confused about their pick up days and Sanitation staff were very accommodating and picked up the items they had by the curb.

Executive Session

- A. Receipt of legal advice relating to claims and potential claims by and against the City and other matters covered by the attorney-client privilege
- B. Discussion of possible contract negotiations for economic development concerning the City of Cayce

Mayor Partin asked the City Attorney if he had anything to discuss under Item VII. A. He stated that he did not. Council Member Corley made a motion to move into Executive Session to discuss Item VII. B. Council Member Almond seconded the motion which was unanimously approved by roll call vote.

Reconvene

After the Executive Session was concluded, Council Member Almond made a motion to reconvene the Regular meeting. Council Member Corley seconded the motion which was unanimously approved by roll call vote. Mayor Partin announced that no vote was taken in Executive Session other than to adjourn and resume the Regular meeting.

Possible Actions by Council in follow up to Executive Session

No action was taken in follow up to Executive Session.

Adjourn

City of Cayce Minutes of 9/18/2019 Council Meeting Page 5

Council Member Carter made a motion to adjourn the meeting. Council Member Almond seconded the motion which was unanimously approved by roll call vote. There being no further business, the meeting adjourned at 5:58 p.m.

	Elise Partin, Mayor	
ATTEST:		
Mendy Corder, CMC, Municipal Clerk		

IF YOU WOULD LIKE TO SPEAK ON A MATTER APPEARING ON THE MEETING AGENDA, PLEASE COMPLETE THE INFORMATION BELOW PRIOR TO THE START OF THE MEETING.* THANK YOU.

COUNCIL MEETING SPEAKERS' LIST

Date of Meeting September 18, 2019

Name	Address	Agenda Item
Robert EVANS -	270/ GRANITE HILL Rd	+

^{*}Appearance of citizens at Council meetings - City of Cayce Code of Ordinances, Sec. 2-71. Any citizen of the municipality may speak at a regular meeting of the council on a matter pertaining to municipal services and operation, with the exception of personnel matters, by notifying the office of the city manager at least five working days prior to the meeting and stating the subject and purpose for speaking. Additionally, during the public comment period as specified on the agenda of a regular meeting of the council, a member of the public may speak on a matter appearing on the meeting agenda, with the exception of personnel matters by signing a speakers list maintained by the city clerk prior to the start of the public comment period. At the discretion of the mayor or presiding officer, the length of time for any speaker's presentation may be limited and the number speakers also may be limited.

Memorandum

To: Mayor and Council

From: Tracy Hegler, City Manager

Blake Bridwell, Director of Utilities

Date: September 26, 2019

Subject: Revised Sewer Use Ordinance and City Ordinance Section 40-92

Issue:

Approval is needed to revise the Sewer Use Ordinance (SUO) by removing the separate Fats, Oils, and Grease (FOG) Control Policy (Attachment A). Approval for a revision to City Ordinance Section 40-92 is also needed for enforcement purposes.

Background/Discussion:

The existing SUO sets forth uniform requirements for users of the Publicly Owned Treatment Works for the City of Cayce and enables the City to comply with all applicable State and Federal laws, including the Clean Water Act (33 United States Code 1251 et seq) and the General Pretreatment Regulations (40 Code of Federal Regulations Part 403).

The SUO has been revised as of November 2018. The revised SUO includes the removal of the FOG Control Policy in the Appendix, however the FOG Control Policy is referenced several times in the revised SUO. The City Attorney and DHEC have approved this revised SUO and DHEC recommends the segregation of the SUO and FOG Control Policy so that the City can adjust the FOG Control Policy without further governmental approvals, when conditions warrant. Governmental approvals are still required for revisions to the SUO.

The FOG Control Policy is to aid staff in the administration of the City of Cayce's Fats, Oils, and Grease Ordinance ("FOG Ordinance") designed to reduce the introduction and accumulation of fats, oils, and grease into the municipal wastewater collection system. This Policy establishes uniform permitting, maintenance and monitoring requirements for controlling the discharge of grease from food service facilities discharging into the wastewater collection system and for regulation of commercial grease haulers operating within the City of Cayce's collection system. This policy was last revised in July 2016. It was a part of the aforementioned SUO, also adopted and approved by City Council at that time.

The FOG Control policy will be a separate document as mentioned above. As a policy, the FOG Control Policy can be changed and modified without further approvals by City Attorney, DHEC, or City Council.

City is requesting a formal legal review of the proposed changes to the SLIO

Recommendation:

Staff recommends Council approve the revised SUO with the removal of the FOG Control Policy. Approval is also recommended for a revision to City Ordinance Section 40-92 per the attached revised language for enforcement purposes.

STATE OF SOUTH CAROLINA) ORDINANCE 2019-16
COUNTY OF LEXINGTON	Amending the Sewer Use Ordinance
CITY OF CAYCE)
technical and updating revisions desires to remove and separate t Grease Control Policy Section") from	es to amend its Sewer Use Ordinance to make certain as indicated by the attached document and further the current Appendix A ("City of Cayce Fats, Oil and om the Sewer Use Ordinance so as to allow it to function y subject to subsequent revisions as a City policy rather
NOW, THEREFORE, BE IT Cayce, in Council, duly assembled	T ORDAINED by the Mayor and Council of the City of d, as follows:
The attached documen adopted as the Sewer U	t, in its final form with indicated changes included, is Jse Ordinance.
of Cayce Fats, Oils and	A of the previous Sewer Use Ordinance entitled "City Grease Control Policy Section" is approved as the City Grease Policy (with deletion of the words "Appendix A" from its title).
This Ordinance shall be effe	ective from the date of second and final reading.
DONE IN MEETING DULY	ASSEMBLED , this day of2019.
Attest:	Elise Partin, Mayor
Mendy Corder, CMC, Municipal C	lerk
First Reading:	
Second Reading and Adoption:	

Danny C. Crowe, City Attorney

Approved as to form:

SEWER USE ORDINANCE CITY OF CAYCE, SC

(Revised July 2016 November 2018)



1800 12TH STREET • CAYCE, SC 29033

P. O. Box 2004 • CAYCE, SC 29171

(803) 796-9020 • FAX: (803) 739-5386

City of Cayce Sewer Use Ordinance **TABLE OF CONTENTS**

TABLE OF CONTENTS	
SECTION 1 - GENERAL PROVISIONS	
1.1 Purpose and Policy	
1.2 Administration.	
1.3 Abbreviations	
1.4 Definitions.	
1.5 Relationship of Ordinance to Wastewater Services Agreements	13
SECTION 2 GENERAL SEWER USE REQUIREMENTS	13
2.1 General Standards	
2.2 General Prohibited Discharge Standards	
2.3 Specific Prohibited Discharge Standards	
2.4 National Categorical Pretreatment Standards	
2.5 State Pretreatment Standards	
2.6 Pretreatment Limits	
2.7 Local Limits	
2.8 Discrepancy between Limits	17
2.9 City's Right of Revision	
2.10 Dilution	18
SECTION 3 - PRETREATMENT OF WASTEWATER	18
3.1 Pretreatment Facilities	18
3.2 Additional Pretreatment Measures	
3.3 Accidental Discharge/Slug Control Plan	19
SECTION 4 - WASTEWATER DISCHARGE PERMIT APPLICATION	20
4.1 Wastewater Analysis	20
4.2 Wastewater Discharge Permit Requirement	
4.3 Wastewater Discharge Permitting: Existing Connections	21
4.4 Wastewater Discharge Permitting: New Connections	21
4.5 Wastewater Discharge Permit Application Contents	22
4.6 Application Signatories and Certification	
4.7 Wastewater Discharge Permit Decisions	23
SECTION 5 - WASTEWATER DISCHARGE PERMIT ISSUANCE PROCESS	23
5.1 Waste Load Allocation	
5.2 Wastewater Discharge Permit Duration	24
5.3 Wastewater Discharge Permit Contents	

ty of Cayce Sewer Use Ordinance	
5.4 Wastewater Discharge Permit Appeals	25
5.5 Wastewater Discharge Permit Modification	26
5.6 Wastewater Discharge Permit Transfer	27
5.7 Wastewater Discharge Permit Revocation	27
5.8 Wastewater Discharge Permit Reissuance	28
5.9 Regulation of Waste Received from Other Jurisdictions	28
SECTION 6 - REPORTING REQUIREMENTS	29
6.1 Baseline Monitoring Reports	
6.2 Compliance Schedule Progress Reports	
6.3 Reports on Compliance with Categorical Pretreatment Standard Deadline	
6.4 Periodic Compliance Reports	
6.5 Reports of Changed Conditions	
6.6 Reports of Potential Problems.	33
6.7 Reports from Unpermitted Users	
6.8 Violation/Repeat Sampling and Reporting	
6.9 Notification of the Discharge of Hazardous Waste	
6.10 Analytical Requirements.	
6.11 Sample Collection	
6.12 Timing	
6.13 Record Keeping	
SECTION 7 - COMPLIANCE MONITORING	
7.1 Right of Entry: Inspection and Sampling	
7.2 Search Warrants	36
SECTION 8 - CONFIDENTIAL INFORMATION	37
SECTION 9 - PUBLICATION OF USERS IN SIGNIFICANT NONCOMPLIANCE	37
SECTION 10 - ADMINISTRATIVE ENFORCEMENT REMEDIES	38
10.1 Notification of Violation	38
10.2 Consent Orders	38
10.3 Show Cause Hearing	39
10.4 Compliance Orders	39
10.5 Cease and Desist Orders	39
10.6 Administrative Fines	40
10.7 Emergency Suspensions	40
10.8 Termination of Discharge	
SECTION 11 - JUDICIAL ENFORCEMENT REMEDIES	12
11.1 Injunctive Relief	
11.1 Injunctive Rener	

11.3 Criminal Prosecution	
11.4 Remedies Nonexclusive	43
SECTION 12 - SUPPLEMENTAL ENFORCEMENT ACTION	43
12.1 Performance Bonds	43
12.2 Liability Insurance	
12.3 Water Supply Severance	
12.4 Public Property Damage	
12.5 Informant Rewards	
12.6 Contractor Listing	
SECTION 13 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS	
13.1 Upset	44
	42 45
13.1 Upset	

SECTION 1 - GENERAL PROVISIONS

1.1 Purpose and Policy

This ordinance sets forth uniform requirements for users of the Publicly Owned Treatment Works for the City of Cayce and enables the City to comply with all applicable State and Federal laws, including the Clean Water Act (33 United States Code § 1251 *et seq.*) and the General Pretreatment Regulations (40 Code of Federal Regulations Part 403). The objectives of this ordinance are:

- +A. To promote and protect the health and safety of the citizens of the City of Cayce;
- 2-B. To prevent the introduction of pollutants into the City of Cayce Publicly Owned Treatment Works that will interfere with its operation;
- 3-C. To prevent the introduction of pollutants into the City of Cayce Publicly Owned Treatment Works that will pass through the Publicly Owned Treatment Works, inadequately treated, into receiving waters, or otherwise be incompatible with the Publicly Owned Treatment Works;
- 4-D. To protect both City of Cayce Publicly Owned Treatment Works personnel who may be affected by wastewater and sludge in the course of their employment and the general public;
- 5-E. To promote reuse and recycling of industrial wastewater and sludge from the City of Cayce Publicly Owned Treatment Works;
- 6-F. To aid in preventing the introduction and accumulation of fats, oils, and greases into the municipal wastewater system which will or tend to cause or contribute to sanitary sewer blockages and obstructions;
- 7-G. To provide for fees for the equitable distribution of the cost of operation, maintenance, and improvement of the City of Cayce Publicly Owned Treatment Works; and
- 8-H. To enable the City to comply with its National Pollutant Discharge Elimination System permit conditions, sludge use and disposal requirements, and any other Federal or State laws to which the City of Cayce Publicly Owned Treatment Works is subject.

This ordinance applies to all users of the City of Cayce Publicly Owned Treatment Works, whether directly or indirectly contributing. The ordinance authorizes enforcement of general requirements for users and issuance of wastewater discharge permits; provides for monitoring, compliance, and enforcement activities; establishes administrative review procedures; requires user reporting; and provides for the setting of fees for the equitable distribution of costs resulting from the program established herein.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

4

1.2 Administration

Except as otherwise provided herein, the Superintendent shall administer, implement, and enforce the provisions of this ordinance. Any powers granted to or duties imposed upon the Superintendent may be delegated by the Superintendent to other City personnel. Any request for a modification or variance to the following requirements of this ordinance shall be in writing and shall provide the information set forth herein along with any information required by the FOG Control Policy.

1.3 Abbreviations

The following abbreviations, when used in this ordinance, shall have the designated meanings:

BMP - Best Management Practice
BOD - Biochemical Oxygen Demand
CFR - Code of Federal Regulations
COD - Chemical Oxygen Demand

EPA - U.S. Environmental Protection Agency

FOG - Fats, Oils and Greases FSE - Food Service Establishment

gpd - gallons per day
IU - Industrial User
mg/l - milligrams per liter

NPDES - National Pollutant Discharge Elimination System

POTW - Publicly Owned Treatment Works RCRA - Resource Conservation and Recovery Act

SCDHEC-South S. Carolina C. Department of Health and Environmental Control

SIC - Standard Industrial Classification

TSS - Total Suspended Solids U.S.C. - United States Code

1.4 Definitions

Unless a provision explicitly states otherwise, the following terms and phrases, as used in this ordinance, shall have the meanings hereinafter designated.

<u>Act or "the Act."</u> The Federal Water Pollution Control Act, also known as the Clean Water Act, as amended, 33 U.S.C. § 1251 et seq.

Approval Authority. The State of South Carolina.

City of Cayce Sewer Use Ordinance Authorized Representative of the User.

- +.(1) If the user is a corporation:
 - (a) The president, secretary, treasurer, or a vice-president of the corporation in charge of a principal business function, or any other person who performs similar policy or decision-making functions for the corporation; or
 - 2-(b) The manager of one or more manufacturing, production, or operation facilities employing more than two hundred fifty (250) persons or having gross annual sales or expenditures exceeding twenty-five (25) million dollars (in second-quarter 1980 dollars), if authority to sign documents has been assigned or delegated to the manager in accordance with corporate procedures.
- 2.(2) If the user is a partnership or sole proprietorship: a general partner or proprietor, respectively.
- 3-(3) If the user is a Federal, State, or local governmental facility: a director or highest official appointed or designated to oversee the operation and performance of the activities of the government facility, or their designee.
- 4-(4) The individuals described in paragraphs 1 through 3, above, may designate another authorized representative if the authorization is in writing, the authorization specifies the individual or position responsible for the overall operation of the facility from which the discharge originates or having overall responsibility for environmental matters for the company, and the written authorization is submitted to the City.

Best Management Practices or BMP. Management and operational procedures that are intended to prevent pollutants from entering a facility's waste stream or from entering a discharge point. BMPs are also schedules of activities, prohibitions of practices, maintenance procedures, and other management practices to prevent or reduce pollution. BMPs also include treatment requirements, operating procedures, and practices to control plant site runoff, spillage or leaks, sludge or waste disposal, or drainage from raw materials storage plus alternative means of complying with certain established categorical Pretreatment Standards and effluent limits as established in the Code of Federal Regulations.

<u>Biochemical Oxygen Demand or BOD</u>. The quantity of oxygen utilized in the biochemical oxidation of organic matter under standard laboratory procedures for five (5) days at <u>twenty degrees Centigrade</u> (20°C)-<u>centigrade</u>, usually expressed as a concentration (e.g., mg/l).

<u>Categorical Pretreatment Standard or Categorical Standard.</u> Any regulation containing pollutant discharge limits promulgated by EPA in accordance with Sections 307(b) and (c) of the Act (33 U.S.C. § 1317) which apply to a specific category of users and which appear in 40 CFR Chapter I, Subchapter N, Parts 405-471.

<u>Categorical User</u>. Any user whose discharge is covered under the categorical pretreatment standard.

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.96", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.96", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.96", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.96", Tab stops: Not at 0.5"

City. The City of Cayce.

<u>Direct Discharge</u>. The discharge of treated or untreated wastewater directly to the waters of the State of South Carolina.

<u>Domestic Waste</u>. Any liquid from bathrooms, shower rooms, toilet rooms, kitchens, laundry facilities located in residences, apartments, hotels, motels, restaurants, cafeterias, office buildings, schools, and commercial establishments. It also includes similar wastes from industries when separated from industrial waste.

Environmental Protection Agency or (EPA). The U.S. Environmental Protection Agency or, where appropriate, the Regional Water Management Division Director, or other duly authorized official of said agency.

Existing Source. Any source of discharge, the construction or operation of which commenced prior to the publication by EPA of proposed categorical pretreatment standards, which will be applicable to such source if the standard is thereafter promulgated in accordance with Section 307 of the Act.

Excessive Foaming. Foaming that can-not be contained within the POTW or adversely affects the operation of it.

<u>Fats, oils, and greases.</u> means oOrganic polar compounds derived from animal and/or plant sources that contain multiple carbon chain triglyceride molecules and non polar compounds (petroleum origin) which are in excess of one hundred (100) mg/l or containing substances which may solidify or become viscous at temperatures between thirty-two (32°F) and one hundred fifty (150) degrees Fahrenheit (150°F) (zero (0°C) and sixty five (65°C) degrees centigrade). These substances are detectable and measurable using analytical test procedures established in 40 CFR 136, as may be amended from time to time. All are sometimes referred to herein as "grease" or "greases" or "FOG".

<u>FOG Control Policy</u>, means t The written plan and procedures by which the superintendent implements and enforces the FOG control and management program established herein. The policy applies to FOG Control Policy violations and matters of program noncompliance. Penalties for specific and programmatic infractions are addressed in the policy and set forth in the city's fee schedule.

<u>Food Service Establishments or "(FSE)"</u>, means those establishments primarily engaged in activities of preparing, serving, or otherwise making available for consumption foodstuffs and that use one or more of the following preparation activities: cooking by frying, baking, grilling, sautéing, rotisserie cooking, broiling, boiling blanching, roasting, toasting, or poaching, and infrared heating, searing, barbecuing, and any other food preparation or serving activity that produces a hot, non-drinkable food product in or on a receptacle requiring washing to be reused and any other facility which in the sole discretion of the Superintendent, must install a Grease Trap or Grease Interceptor prior to discharging kitchen or food preparation wastewater into the City of Cayce's wastewater collection system.

Food Truck. A fully enclosed mobile kitchen that may be used to prepare, cook, or serve time/temperature control for food safety.

<u>Grab Sample</u>. A sample which is taken from a wastestream without regard to the flow in the wastestream and over a period of time not to exceed fifteen (15) minutes.

<u>Gray Water</u>. means aAll the liquid contained in a Grease trap or Grease Interceptor that lies below the floating grease layer and above the bottom solids.

<u>Grease Hauler</u>. means aAny a person or entity that collects the contents of Grease Traps and/or Grease Interceptors and transports the contents to an approved recycling or disposal facility. A Grease Hauler may also provide other services to FSE's related to Grease Trap and/or Grease Interceptor cleaning.

<u>Grease Trap.</u> is a A device located within an FSE designed to collect, contain, separate, or remove Grease from the waste stream while allowing the sub-straight waste or Gray Water to discharge to the wastewater collection system by gravity. Also known as a grease removal device (GRD).

<u>Grease Interceptor</u>: is a vault located outside or inside of an FSE or IU designed to collect, contain, separate, or remove grease, sand fats, wax, flammable liquids, oils of hydrocarbon or petroleum origin and substances which may solidify or become viscous in the system from the waste stream while allowing the sub-straight waste or Gray Water to discharge to the wastewater collection system by gravity. Also known as a grease removal device (GRD).

<u>Headworks Analysis</u>. Determines the maximum amounts of various pollutants that could be discharged at the headworks of the POTW without violating the receiving stream limits. This analysis is based on the minimum flow of the receiving stream, discharges from other POTW's to the receiving stream, and contributions from all users to the City POTW. The Head Works Analysis is updated from time to time.

<u>Indirect Discharge or Discharge</u>. The introduction of pollutants into the POTW from any domestic or non-domestic source regulated under section 307(b), (c) or (d) of the Act

<u>Industrial User or "(IU)"</u>. A source of indirect discharge which does not constitute a "discharge of pollutants" under regulations issued pursuant to Section 402, of the Federal Water Pollution Control Act (33 U.S.C. 1342).

Industrial Waste. Liquid waste from industrial processes.

<u>Instantaneous Maximum Allowable Discharge Limit</u>. The maximum concentration of a pollutant allowed to be discharged at any time, determined from the analysis of any discrete or composited sample collected, independent of the industrial flow rate and the duration of the sampling event.

<u>Interference</u>. A discharge, which alone or in conjunction with a discharge or discharges from other sources, inhibits or disrupts the POTW, its treatment processes or operations or its sludge processes, use or disposal; and including an increase in the magnitude or duration of a violation of the City's NPDES permit or of the prevention of sewage sludge use or disposal in compliance with any of the following statutory/regulatory provisions or permits issued thereunder, or any more stringent State or local regulations: Section 405 of the Act; the Solid Waste Disposal Act, including Title II commonly referred to as the Resource Conservation and Recovery Act (RCRA); any State regulations contained in any State sludge management plan prepared pursuant to Subtitle D of the Solid Waste Disposal Act; the Clean Air Act; the Toxic Substances Control Act; and the Marine Protection, Research, and Sanctuaries Act.

May. Is permissive.

<u>Medical Waste</u>. Isolation wastes, infectious agents, human blood and blood products, pathological wastes, sharps, body parts, contaminated bedding, surgical wastes, potentially contaminated laboratory wastes, and dialysis wastes.

Minimum design capability. is-tThe design features of a grease interceptor and its ability or volume required to effectively intercept and retain grease and settled solids from grease-laden wastewater prior to discharge to the public sanitary sewer.

<u>Net Adjustment</u>. Adjustment made to the user's Pretreatment Standard(s), based on the concentration of pollutant(s) in the user's intake water pursuant to Part 403.15 of 40 CFR.

New Source.

- 1-(1) Any building, structure, facility, or installation from which there is (or may be) a discharge of pollutants, the construction of which commenced after the publication of proposed pretreatment standards under Section 307(c) of the Act which will be applicable to such source if such standards are thereafter promulgated in accordance with that section, provided that:
 - +(a) The building, structure, facility, or installation is constructed at a site at which no other source is located; or
 - 2-(b) The building, structure, facility, or installation totally replaces the process or production equipment that causes the discharge of pollutants at an existing source; or
 - 3-(c) The production or wastewater generating processes of the building, structure, facility, or installation are substantially independent of an existing source at the same site. In determining whether these are substantially independent, factors such as the extent to which the new facility is integrated with the existing plant, and the extent to which the new facility is engaged in the same general type of activity as the existing source, should be considered.
- 2-(2) Construction on a site at which an existing source is located results in a modification rather than a new source if the construction does not create a new building, structure, facility, or installation meeting the criteria of Section (1)(b) or (c) above but otherwise alters, replaces, or adds to existing process or production equipment.
- 3-(3) Construction of a new source as defined under this paragraph has commenced if the owner or operator has:
 - 4.(d) Begun, or caused to begin, as part of a continuous on-site construction program
 - +.(i) any placement, assembly, or installation of facilities or equipment; or
 - 2-(ii) significant site preparation work including clearing, excavation, or removal of

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.9", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.96" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.96" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.96" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.9", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.65" + Indent at: 0.9", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.96" + Indent at: 1.21", Tab stops: Not at 1"

Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Left + Aligned at: 1.15" + Indent at: 1.53", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: i, ii, iii, ... + Start at: 1 + Alignment: Left + Aligned at: 1.15" + Indent at: 1.53", Tab stops: Not at 1.5"

9

existing buildings, structures, or facilities which is necessary for the placement, assembly, or installation of new source facilities or equipment; or

2-(e) Entered into a binding contractual obligation for the purchase of facilities or equipment which are intended to be used in its operation within a reasonable time. Options to purchase or contracts which can be terminated or modified without substantial loss, and contracts for feasibility, engineering, and design studies do not constitute a contractual obligation under this paragraph.

<u>Noncontact Cooling Water</u>. Water used for cooling which does not come into direct contact with any raw material, intermediate product, waste product, or finished product.

<u>Pass Through</u>. A discharge which exits the POTW into waters of the United States in quantities or concentrations which, alone or in conjunction with a discharge or discharges from other sources, is a cause for violation of any requirement of the City's NPDES permit, including an increase in the magnitude or duration of a violation.

<u>Person</u>. Any individual, partnership, copartnership, firm, company, corporation, association, joint stock company, trust, estate, governmental entity, or any other legal entity; or their legal representatives, agents, or assigns. This definition includes all Federal, State, and local governmental entities.

<u>Permit.</u> is a A document that confirms approval issued by the Superintendent. The User is required to keep the Permit on the premises and produce it upon request of the Superintendent.

<u>pH</u>. Logarithm of the reciprocal of the <u>weight_concentration</u> of hydrogen ions in moles (<u>gram moles</u>) per liter (<u>mol/L</u>) of solution.

<u>Pollutant</u>. Substances that include, but are not limited to, Ddredged spoil, solid waste, incinerator residue, filter backwash, sewage, garbage, sewage sludge, munitions, medical wastes, chemical wastes, biological materials, radioactive materials, heat, wrecked or discarded equipment, rock, sand, cellar dirt, municipal, agricultural and industrial wastes, and certain characteristics of wastewater (e.g., pH, temperature, TSS, turbidity, color, BOD, COD, toxicity, or odor), and any other parameter that the superintendent interprets as having an effect on the POTW.

<u>Pollution</u>. The man-made or man-induced alteration of the chemical, physical, biological, and radiological integrity of water.

<u>Pretreatment</u>. The reduction of the amount of pollutants, the elimination of pollutants, or the alteration of the nature of pollutant properties in wastewater prior to, or in lieu of, introducing such pollutants into the POTW. This reduction or alteration can be obtained by physical, chemical, or biological processes; by process changes; or by other means, except by diluting the concentration of the pollutants unless allowed by an applicable pretreatment standard.

<u>Pretreatment Program</u>. A set of procedures designed to help achieve the City the objective listed in Section 1.1 of this document.

 $\underline{\textit{Pretreatment Requirements}}$. Any substantive or procedural requirement related to pretreatment imposed on a user, other than a pretreatment standard.

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.96" + Indent at: 1.21", Tab stops: Not at 1"

<u>Pretreatment Standards or Standards</u>. Pretreatment standards shall mean prohibited discharge standards, categorical pretreatment standards, and local limits.

<u>Prohibited Discharge Standards or Prohibited Discharges</u>. Absolute prohibitions against the discharge of certain substances; these prohibitions appear in Section 2.2 of this ordinance.

<u>Publicly Owned Treatment Works or (POTW)</u>. A "treatment works," as defined by Section 212 of the Act (33 U.S.C. §1292) which is owned by the City. This definition includes any devices or systems used in the collection, storage, treatment, recycling, and reclamation of sewage or industrial wastes of a liquid nature and any conveyances which convey wastewater to a treatment plant.

<u>Receiving Stream</u>. Is that body of water, stream, or watercourse receiving the discharge from a waste treatment plant, or that body of water, stream, or watercourse formed by the effluent from a waste treatment plant.

<u>Recurring Violation</u>. A minimum of three (3) violations of a similar nature in any six (6) consecutive month period.

<u>Septic Tank Waste</u>. Any sewage from holding tanks such as vessels, chemical toilets, campers, trailers, and septic tanks.

<u>Sewage or wastewater</u>. Shall mean a combination of the water-carried wastes from residences, business buildings, institutions and industrial establishments together with any groundwater, surface water, and storm water that may be present.

Shall. Is mandatory.

Significant Industrial User.

- 4.(1) A user subject to categorical pretreatment standards; or
- 2.(2) A user that:
 - 1-(a) Discharges an average of twenty-five thousand (25,000) gpd or more of process wastewater to the POTW (excluding sanitary, noncontact cooling, and boiler blowdown wastewater);
 - 2-(b) Contributes a process waste stream which makes up five (5) percent or more of the average dry weather hydraulic or organic capacity of the POTW treatment plant; or
 - 3-(c) Is designated as such by the City on the basis that it has a reasonable potential for adversely affecting the POTW's operation or for violating any pretreatment standard or requirement.
- 3-(3) Upon a finding that a user meeting the criteria in Subsection (2) has no reasonable potential for adversely affecting the POTW's operation or for violating any pretreatment standard or requirement, the City may at any time, on its own initiative or in response to a petition

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.71" + Indent at: 0.96", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.71" + Indent at: 0.96", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.71" + Indent at: 0.96", Tab stops: Not at 0.5"

11

received from a user, and in accordance with procedures in 40 CFR 403.8(f)(6), determine that such user should not be considered a significant industrial user.

<u>Significant Noncompliance</u>. Criteria used to identify important violations and/or patterns of noncompliance with the discharge limits or other provisions of the City Sewer Use Ordinance as defined in Section 9 of this ordinance.

<u>Slug Load or Slug</u>. Any discharge at a flow rate or concentration which could cause a violation of the prohibited discharge standards in Section 2.2 of this ordinance.

<u>Standard Industrial Classification (SIC) Code.</u> A classification pursuant to the <u>Standard Industrial Classification Manual</u> issued by the United States Office of Management and Budget.

<u>Storm Water</u>. Any flow occurring during or following any form of natural precipitation, and resulting from such precipitation, including snowmelt.

<u>Superintendent</u>. The person designated by the City to supervise the operation of the POTW, and who is charged with certain duties and responsibilities by this ordinance, or a duly authorized representative.

<u>Suspended Solids</u>. The total suspended matter that floats on the surface of, or is suspended in, water, wastewater, or other liquid, and which is removable by laboratory filtering.

<u>Toxic Pollutant</u>. Any pollutant or combination of pollutants listed astoxic in regulations promulgated by the Administrator of the Environmental Protection Agency under the provisions of CWA(a) or other Clean Water Acts.

<u>Toxicity Test</u>. A procedure approved by SCDHEC, in which the responses of aquatic organisms are used to measure or detect the presence or effects of one or more substances, wastes, or environmental factors, alone or in combination.

User.

1.(1) A source of indirect discharge.

2-(2) Is the responsible person or entity for the FSE operations or in the case of the individual FSE, the owner or proprietor of the food service operation. Where the FSE is a franchise operation, the owner of the franchise is the responsible person or entity. Where the FSE operation is owned by a corporation, the corporate representative is the responsible entity. Where two or more FSE share a common grease interceptor, the User shall be the individual who owns or assumes control of the grease interceptor or the property on which the grease interceptor is located.

<u>Utility Rate Schedule</u>. A document updated from time to time which establishes the exact amounts of all categories of fees based upon various criterion such as meter sizes, flow, or other suitable criteria for the fee identified.

<u>Waste Load Allocation</u>. Is the establishment of flows, and pollutant limits and loads for the proposed non-domestic discharges based on the POTW Head Works Analysis which is updated from time to time.

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.81", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.5" + Indent at: 0.81", Tab stops: Not at 0.5"

<u>Wastewater</u>. Liquid and water-carried industrial wastes and sewage from residential dwellings, commercial buildings, industrial and manufacturing facilities, and institutions, whether treated or untreated, which are contributed to the POTW.

<u>Wastewater Collection System</u>. That portion of the POTW which collects and conveys wastewater to the wastewater treatment plant, which consists of gravity or force mains, pumps, and other appurtenances.

<u>Wastewater Treatment Plant or Treatment Plant</u>. That portion of the POTW which is designed to provide treatment of municipal sewage and industrial waste.

1.5 Relationship of Ordinance to Wastewater Services Agreements

As to a User which also is a "Purchaser" within the meaning of the Wastewater Services Agreement among the City of Cayce, Town of Lexington and the Joint Municipal Water and Sewer Commission dated August 28, 2009, as amended ("the 2009 Agreement"), the terms and provisions of the 2009 Agreement and the Purchaser's Wastewater Contribution Permit shall control over any conflicting terms and provisions of this ordinance. The City of Cayce acknowledges that the Town of Lexington and the JMW&SC each have a FOG program in place and are administering and overseeing these FOG Programs. Nothing in this Sewer Use Ordinance, or its exhibits hereto, shall impose additional requirements relating to these FOG programs of the Town of Lexington or JMW&SC, including, but not limited to, the provisions of paragraph 3.2 below.

SECTION 2 GENERAL SEWER USE REQUIREMENTS

2.1 General Standards

+A. Within City limits, if sewer service is available within a distance of two hundred and fifty (250) feet from the property boundary, the property owner shall tie-on to the City sewer line. In case of an existing building, the property owner shall initiate sewer service application process within one hundred and eighty (180) days from the date of sewer availability or the adoption of this ordinance, whichever is later, according to Section 2.1 B of this ordinance, provided that the existing septic tank is in working order. All newly constructed buildings shall be connected to the City sewer, if sewer service is available prior to the issuance of a certificate of occupancy. In the event that an owner does not tie on to sewer service, if available, the City shall charge for sewer service as if the service is being used pursuant to Sections 14-and 15. Property owners who, as of December 1, 2011, are known to the City not to be connected to, and discharging sewer into, the sewer system shall have the option to elect, in writing, to discontinue payment of the availability charge described in this subsection A but, upon exercise of that option, and any subsequent connection to the sewer system, shall be subject to payment of the then current fees described in subsection B. Property owners who do not discontinue payment of the sewer availability charge described in this subsection A shall be entitled, upon any subsequent connection to the sewer system, to a credit of the amount of subsection A charges paid against, but not to exceed, the amount of any subsection

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

B fees due. Sewer service using privies, cesspools, and/or septic tanks is prohibited in the areas serviced by the City sewer. Existing septic tanks may not be repaired, modified, pumped out or in any way maintained in areas serviced by City of Cayce sewer. Septic tank owners will be required to initiate sewer service application process immediately upon expiration of septic tank capacity provided sewer service is available.

- 2-B. Any person desiring to connect with the City's sanitary sewer system shall obtain a permit for such connection upon paying applicable fees and according to the procedure outlined in Section 4.4 of this ordinance.
- 3-C. Every building having plumbing fixtures installed intended for human habitation, occupancy, or use on premises abutting on a street, alley, or easement in which there is a public sewer shall have a separate connection with the City sewer, except by special permits in accordance with City of Cayce Code 40-99 adopting Southern Building Code. This ordinance is in Section 701.2 of Southern Building Code. Reuse of building sewers is not permitted, unless approved by the Superintendent.
- 4-D. Building sewer line work shall not be commenced until the plans and specifications have been approved by City of Cayce. Such work shall be performed under the supervision and to the satisfaction of the City of Cayce as outlined in City of Cayce Code Section 40-98. All building sewer line materials, grades, and sizes shall be according to the City of Cayce Code Section 40-99, which adopts the Southern Building Code. This ordinance is covered by Standard Plumbing Code Sections 7 Sanitary Drainage. Sewer line construction or maintenance work shall follow the Occupational Safety and Health Administration Safety Standards as outlined in 29 CFR 1910.
- 5-E. If a user located within another municipal jurisdiction contributes wastewater to the City of Cayce POTW, the user's building sewer work shall be performed according to Section 2.1D of this ordinance. Failure to comply with this provision will result in refusal of sewer service by the City of Cayce.
- 6-F. Septic tank services or other maintenance services may not contract for the removal of septic tank sludge or any maintenance of septic tank on any property serviced by City sewer. Any company or firm which violates this provision shall be subjected to the same fines and other enforcement actions that are applicable to the user or septic tank owner.

2.2 General Prohibited Discharge Standards

These general prohibitions apply to all users of the POTW whether or not they are subject to categorical pretreatment standards or any other National, State, or local pretreatment standards or requirements:

- 4-A. No person shall discharge septic tank effluent or cesspool overflow to any open drain, ditch, stream, or well penetrating waterbearing formations, if cesspools and septic tanks are allowed in areas where no sewer service is available.
- 2.B. No user shall introduce or cause to be introduced into the POTW:
 - 4.1) Any pollutant or wastewater which causes pass through or interference;

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 3 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

- 2-2) Any pollutant or wastewater to which water is added for the purpose of diluting wastes that would otherwise exceed applicable maximum concentration limitations for any wastewater constituent; or
- 3-3 Any substance which will cause the POTW to violate its NPDES permit or the receiving water quality standards.

2.3 Specific Prohibited Discharge Standards

The following are specific prohibitions. No user shall introduce or cause to be introduced into the POTW the following pollutants, substances, or wastewater:

- 1-A. Pollutants which create a fire or explosive hazard in the POTW, including, but not limited to, wastestreams with a closed-cup flashpoint of less than 140°F (60°C) using the test methods specified in 40 CFR 261.21;
- 2-B. Pollutants which will cause corrosive structural damage to the POTW, but in no case discharges with lower than 5.0 or more than 10.0, unless the works is specifically designed to accommodate such discharges;
- 3-C. Solid or viscous substances in amounts which will cause obstruction of the flow in the POTW resulting in interference but in no case solids greater than one-half inch (1/2") or 1.25 centimeter (1.25 cm) in any dimension;
- 4-D. Pollutants, including oxygen-demanding pollutants (BOD, etc.), released in a discharge at a flow rate and/or pollutant concentration which, either singly or by interaction with other pollutants, will cause interference with the POTW;
- 5-E. Wastewater having a temperature greater than 150140°F (6560°C), or which will inhibit biological activity in the treatment plant resulting in interference, but in no case wastewater which causes the temperature at the introduction into the treatment plant to exceed 104°F (40°C);
- 6-F. Petroleum oil, nonbiodegradable cutting oil, or products of mineral oil origin, in amounts that will cause interference or pass through;
- 7.G. Pollutants which result in the presence of toxic gases, vapors, or fumes within the POTW in a quantity that may cause acute worker health and safety problems;
- 8-H. Trucked or hauled pollutants, except at discharge points designated by the Superintendent in accordance with Section 3.4 of this ordinance The City of Cayce's FOG Control Policy;
- D.I. Noxious or malodorous liquids, gases, solids, or other wastewater which, either singly or by interaction with other wastes, are sufficient to create a public nuisance or a hazard to life, or to prevent entry into the sewers for maintenance or repair;
- 10.1 Wastewater which imparts color which cannot be removed by the treatment process at the POTW, such as, but not limited to, dye wastes and vegetable tanning solutions;

Formatted: Numbered + Level: 3 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

- 41.K. Wastewater containing any radioactive wastes or isotopes except in compliance with applicable State or Federal regulations;
- Storm water, surface water, groundwater, artesian well water, roof runoff, subsurface drainage, swimming pool drainage, condensate, deionized water, noncontact cooling water, and other unpolluted discharges, unless specifically authorized by the Superintendent;
- 43.M. Sludges, screenings, or other residues from the pretreatment of industrial wastes;
- 14.N. Medical wastes, except as specifically authorized by the Superintendent in a wastewater discharge permit;
- Use Mastewater causing, alone or in conjunction with other sources, the treatment plant's effluent to fail a toxicity test;
- 6.P. Detergents, surface-active agents, or other substances which may cause excessive foaming in the POTW;
- 47.0. Fats, oils, or greases of animal or vegetable origin in concentrations greater than limits established in Section 2.45 and 2.6;
- Wastewater causing two readings on an explosion hazard meter at the point of discharge into the POTW, or at any point in the POTW, of more than five percent (5%) or any single reading over ten percent (10%) of the Lower Explosive Limit of the meter.
- 19.S. Discharge of sanitary wastewater into the storm sewer system without exception.

Pollutants, substances, or wastewater prohibited by this section shall not be processed or stored in such a manner that they could be discharged to the POTW.

2.4 National Categorical Pretreatment Standards

The categorical pretreatment standards found at 40 CFR Chapter I, Subchapter N, Parts 405-471 are hereby incorporated.

- 4-A. Where a categorical pretreatment standard is expressed only in terms of either the mass or the concentration of a pollutant in wastewater, the Superintendent may impose equivalent concentration or mass limits in accordance with R.61-9.403.6(d)(2).
- 2-B. When wastewater subject to a categorical pretreatment standard is mixed with wastewater not regulated by the same standard, the Superintendent shall impose an alternate limit using the combined wastestream formula in R.61-9.403.6(f).
- 3.C. A user may obtain a variance from a categorical pretreatment standard if the user can prove, pursuant to the procedural and substantive provisions in R.61-9.403.13, that factors relating to its

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

discharge are fundamentally different from the factors considered by EPA when developing the categorical pretreatment standard.

4-D. A user may obtain a net adjustment to a categorical standard in accordance with R.61-9.403.15.

2.5 State Pretreatment Standards

Any applicable pretreatment standards as adopted by the State of South Carolina are hereby incorporated as if fully listed herein.

2.6 Pretreatment Limits

Pollutant limits are established to protect against pass through and interference, and to comply with State and Federal Regulations. No person user shall discharge wastewater containing in excess of the maximum allowable discharge limits as identified in the most current Industrial User wastewater discharge contribution permits.

These limits apply at the point where the wastewater is discharged to the POTW. All concentrations for metallic substances are for "total" metal unless indicated otherwise. The Superintendent may impose mass limitations for a few or all pollutants, in addition to, or in place of, the concentration-based limitations listed in the Headworks Analysis in accordance with the pretreatment program.

2.7 Local Limits

- A. The Superintendent is authorized to establish Local Limits pursuant to 40 CFR 403.5(c).
- B. Any local limits established are to protect against Pass Through and Interference. No person shall discharge wastewater containing in excess of the allowable local limits.

2.8 Discrepancy between Limits

In the event that a discrepancy occurs between one or more of the discharge limits as required by the National Categorical Pretreatment Standards, State Pretreatment Standards, or Local Pretreatment Limits, or the Local Limits, then the most stringent of the standards will govern.

2.9 City's Right of Revision

The City reserves the right to establish, by ordinance, more stringent standards or requirements on

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.25" + Indent at: 0.5", Tab stops: Not at 0.5"

discharges to the POTW. The City shall also have the right to establish, in wastewater discharge permits, more stringent standards or requirements on discharges to the POTW:

- +A. If the Superintendent feels that such discharges adversely affect the operation of the POTW;
- 2-B. Based on the capacity of the POTW either hydraulically or pollutant loads as determined by the most recent Headworks Analysis; or
- 3.C. To provide an equitable share to all users or for future anticipated discharges to the POTW.

2.10 Dilution

No user shall increase the use of process water, or in any other way dilute or attempt to dilute a discharge, as a partial or complete substitute for adequate treatment to achieve compliance with a discharge limitation unless expressly authorized by an applicable pretreatment standard or requirement. The Superintendent may impose mass limitations on users who are using dilution to meet applicable pretreatment standards or requirements, or in other cases when the imposition of mass limitations is appropriate.

SECTION 3 - PRETREATMENT OF WASTEWATER

3.1 Pretreatment Facilities

Users shall provide wastewater treatment as necessary to comply with this ordinance and shall achieve compliance with all categorical pretreatment standards, local limits, and the prohibitions set out in Section 2.2 of this ordinance within the time limitations specified by the State or the Superintendent, whichever is more stringent. All facilities necessary for compliance shall be provided, operated, and maintained at the user's expense. Detailed plans describing such facilities and operating procedures shall be submitted to the Superintendent for review, and shall be approved by the Superintendent before such facilities are constructed. The review of such plans and operating procedures shall in no way relieve the user from the responsibility of modifying such facilities as necessary to produce a discharge in compliance with the provisions of this ordinance or obtaining permits to construct or operate from other agencies or South Carolina Department of Health and Environmental Control (SCDHEC).

3.2 Additional Pretreatment Measures

Whenever deemed necessary for protection or preservation of public health and safety and the POTW, the Superintendent may require users to restrict their discharge during peak flow

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

periods, designate that certain wastewater be discharged only into specific sewers, relocate and/or consolidate points of discharge, separate sewage wastestreams from industrial wastestreams, and impose such other conditions as may be necessary to protect the POTW and determine the user's compliance with the requirements of this ordinance.

- 2-B. The Superintendent may require any person discharging into the POTW to install and maintain, on their property and at their expense, a suitable storage and flow-control facility to ensure equalization of flow. A wastewater discharge permit may be issued solely for flow equalization.
- 3.C. Grease, oil, and sand interceptors shall be provided utilized for the proper handling of wastewater containing excessive amounts of grease and oil, which will or tend to cause or contribute to sanitary sewer blockages and obstructions, or sand; except that such interceptors shall not be required for residential users. Food Service Establishments, commercial establishments and other industrial generating wastewater containing fats, oils, or greases are subject to the City of Cayce's Fats, Oils, and Grease Control Policy and this ordinance.

3.3 Accidental Discharge/Slug Control Plan

The Superintendent shall evaluate whether each significant industrial user needs an accidentally discharge/slug control plan. The Superintendent may require any User to develop, submit for approval, and implement such a plan or take such other action that may be necessary to control Slug Discharges. Alternatively, the Superintendent may develop such a plan for any user. An accidental discharge/slug discharge control plan shall address, at a minimum, the following:

- 4.A. Description of discharge practices, including nonroutine batch discharges;
- 2.B. Description of stored chemicals;
- 3-C. Procedures for immediately notifying the Superintendent of any accidental or slug discharge and any facility changes affecting the potential for a slug discharge as required by Section 6.6 of this ordinance; and
- 4-D. Procedures to prevent adverse impact from any accidental or slug discharge. Such procedures include, but are not limited to, inspection and maintenance of storage areas, handling and transfer of materials, loading and unloading operations, control of plant site runoff, worker training, building of containment structures or equipment, measures for containing toxic organic pollutants, including solvents, and/or measures and equipment for emergency response.

3.4 Hauled Wastewater

The following regulations shall apply to haulers of septic and industrial wastewater:

LA. Septic tank waste may be introduced into the POTW only at locations designated by the Superintendent, and at such times as are established by the Superintendent. Such waste shall not violate Section 2 of this ordinance or any other requirements established by the City. The

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.52", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.52", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Superintendent shall require septic tank waste haulers to obtain wastewater discharge permits. All such discharges are subject to the appropriate fees based on out-of-town rates as described in Sections 14 and 15 of this ordinance.

- 2-B. The Superintendent shall require haulers of industrial waste to obtain wastewater discharge permits. The Superintendent may require generators of hauled industrial waste to obtain wastewater discharge permits. The Superintendent also may prohibit the disposal of hauled industrial waste. The discharge of hauled industrial waste is subject to all other requirements of this ordinance. All such discharges are subject to the appropriate fees based on out-of-town rates as described in Sections 14 and 15 of this ordinance.
- No load may be discharged without prior consent of the Superintendent. The Superintendent may collect samples of each hauled load to ensure compliance with applicable standards. The Superintendent may require the industrial waste hauler to provide a waste analysis of any load prior to discharge. Any industrial hauler who misrepresents the characteristics of the waste being discharged shall be subjected to immediate permit revocation, in addition to enforcement response actions according to Sections 10 and 11 of this ordinance.
- 4-D. Industrial waste haulers must provide a waste-tracking form for every load. This form shall include, at a minimum, the name and address of the industrial waste hauler, permit number, truck identification, names and addresses of sources of waste, and volume and characteristics of waste. The form shall identify the type of industry, known or suspected waste constituents, and whether any wastes are RCRA hazardous wastes.

SECTION 4 - WASTEWATER DISCHARGE PERMIT APPLICATION

4.1 Wastewater Analysis

When requested by the Superintendent, a user must submit information on the nature and characteristics of its wastewater within fifteen (15) days of the request. The Superintendent is authorized to prepare a form for this purpose and may periodically require users to update this information. Failure by a user to submit the required information may result in fines or termination of service as determined by the Superintendent.

4.2 Wastewater Discharge Permit Requirement

- +A. No significant industrial user shall discharge wastewater into the POTW without first obtaining a wastewater discharge permit from the Superintendent, except that a significant industrial user that has filed a timely application pursuant to Section 4.3 of this ordinance may continue to discharge for the time period specified therein.
- 2-B. The Superintendent may require other users to obtain wastewater discharge permits as necessary to carry out the purposes of this ordinance.

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.28" + Indent at: 0.53", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.28" + Indent at: 0.53", Tab stops: Not at 0.5"

Formatted: Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.28" + Indent at: 0.53", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

3.C. Any violation of the terms and conditions of a wastewater discharge permit shall be deemed a violation of this ordinance and subject the wastewater discharge permittee to the sanctions set out in Sections 10 through 12 of this ordinance. Obtaining a wastewater discharge permit does not relieve a permittee of its obligation to comply with all Federal and State pretreatment standards or requirements or with any other requirements of Federal, State, and local law.

4.3 Wastewater Discharge Permitting: Existing Connections

Any user required to obtain a wastewater discharge permit who was discharging wastewater without a permit into the City POTW prior to the effective date of this ordinance and who wishes to continue such discharges in the future, shall, within thirty (30) days after said date, apply to the Superintendent for a wastewater discharge permit in accordance with Section 4.5 of this ordinance, and shall not cause or allow discharges to the POTW to continue after forty five (45) days of the effective date of this ordinance except in accordance with a wastewater discharge permit issued by the Superintendent.

4.4 Wastewater Discharge Permitting: New Connections

Anyone who proposes to begin discharging into the POTW must obtain a permit prior to the beginning or recommencing of such discharge. The user shall fill out a form identifying the type of discharge, domestic or non-domestic. The user identifying the proposed discharge as domestic, shall certify as follows:

"I certify under penalty of law that this document and all attachments were prepared under my direction or supervision in accordance with a system designed to assure that qualified personnel properly gather and evaluate the information submitted. Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to the best of my knowledge and belief, true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing violations."

The user identifying the proposed discharge as domestic, shall be allowed to connect to the City sewer upon payment of the applicable fees defined in Section 14 of this Ordinance and in the most recent edition of City of Cayce-Utility Rate schedule. These fees shall include, but not limited to Tap Fee and the Impact Fee (Capacity Charge).

The user identifying the proposed discharge as non-domestic, shall fill out a waste survey questionnaire and an application form in accordance with Section 4.5 of this ordinance. All applications for wastewater discharges, whether domestic or non-domestic, must be filed at least ninety (90) days prior to the date upon which any discharge will begin.

4.5 Wastewater Discharge Permit Application Contents

All users identifying the proposed discharge as non-domestic must fill out a waste survey questionnaire and submit a permit application. The Superintendent may require all users to submit as part of an application the following information:

- 1.A. All information required by Section 6.1(B) of this ordinance;
- 2-B. Description of activities, facilities, and plant processes on the premises, including a list of allraw materials and chemicals used or stored at the facility which are, or could accidentally or intentionally be, discharged to the POTW;
- 3-C. Number and type of employees, hours of operation, and proposed or actual hours of operation;
- 4.D. Each product produced by type, amount, process or processes, and rate of production;
- 5.E. Type and amount of raw materials processed (average and maximum per day);
- 6-F. Site plans, floor plans, mechanical and plumbing plans, and details to show all sewers, floor drains, and appurtenances by size, location, and elevation, and all points of discharge;
- 7.G. Time and duration of discharges; and
- 8-H. Any other information as may be deemed necessary by the Superintendent to evaluate the wastewater discharge permit application.

Incomplete or inaccurate applications will not be processed and will be returned to the user for revision. The Superintendent may revoke or modify a wastewater discharge permit according to Sections 5.4 and 5.6 of this ordinance.

4.6 Application Signatories and Certification

4-A. All non-domestic wastewater discharge permit applications and user reports must be signed by an authorized representative of the user and contain the following certification statement:

"I certify under penalty of law that this document and all attachments were prepared under my direction or supervision in accordance with a system designed to assure that qualified personnel properly gather and evaluate the information submitted. Based on my inquiry of the person or persons who manage the system, or those persons directly responsible for gathering the information, the information submitted is, to the best of my knowledge and belief, true, accurate, and complete. I am aware that there are significant penalties for submitting false information, including the possibility of fine and imprisonment for knowing Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

City of Cayce Sewer Use Ordinance violations."

2-B. If the designation of an Authorized Representative is no longer accurate because a different individual or position has responsibility for the overall operation of the facility or overall responsibility for environmental matters for the company, a new written authorization satisfying the requirements of Section 1.4 C must be submitted to the Superintendent prior to or together with any reports to be signed by an Authorized Representative. Letters of Signatory Authority are to be submitted annually.

4.7 Wastewater Discharge Permit Decisions

A wastewater discharge permit application must be submitted one hundred eighty (180) days before expected discharge. Within forty five (45) days of Upon receipt of a complete application waste survey questionnaire furnished by the non-domestic user, the Superintendent will evaluate the data and determine whether or not to issue a wastewater discharge permit. The Superintendent may deny any application for a wastewater discharge permit for the following reasons:

- 4.A. Failure to furnish complete information as required in Sections 4.5 and 5.1 of this ordinance;
- 2.B. Inadequate or inappropriate pretreatment proposed;
- 3.C. Potential for discharge of toxic chemical(s);
- 4.D. Failure to pay the appropriate fees;
- 5.E. For falsifying or misrepresentation of proposed discharges or other information; and
- 6-F. For any other reason(s) the Superintendent determines that the proposed discharges will adversely affect the operation of the POTW or the implementation of the local pretreatment program.

SECTION 5 - WASTEWATER DISCHARGE PERMIT ISSUANCE PROCESS

5.1 Waste Load Allocation

The Superintendent will evaluate the waste survey questionnairewastewater discharge permit application data furnished by the non-domestic user and may require additional information. Within forty five (45) days of Upon receipt of a complete applicationwaste survey questionnaire and other information, the Superintendent will establish the waste load allocation limits for the proposed discharge. Upon payment of a non-refundable waste load allocation fee the Superintendent will issue a Waste Load Allocation, setting the discharge capacity and limits in the City POTW for the proposed discharges by the non-domestic user. The Waste Load Allocation will be issued for a period of one (1) year; or, until such time as the non-domestic user submits the final plans or sufficient documentation to

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stoos: Not at 0.5"

the City and a permit is issued by the City, whichever comes first. The Waste Load Allocation is not transferable to any other user or for any other discharge than the proposed discharge. Payment of the waste allocation fee will secure the non-domestic user's capacity and limits in the City POTW as set forth in the Waste Load Allocation for a period of one (1) year, provided that suitable pretreatment facilities are proposed or sufficient documents are provided by the non-domestic user that are acceptable to the Superintendent and that will demonstrate to a reasonable certainty that the limits can be met. Approval of plans or documentation by the Superintendent does not relieve the non-domestic user from the liability of the design of the pretreatment facilities nor full compliance with any permits issued. Within one (1) year from the date of establishment of waste load allocation limits, the non-domestic user shall submit to the City, an updated and completed application form that matches any design numbers and criteria developed for DHEC permitting, final plans and other information as discussed under Section 4.5 of this ordinance. Failure to submit this information will result in the expiration of the Waste Load Allocation without further remedy unless a prior written request by the non-domestic user for an extension of the Waste Load Allocation has been received by the Superintendent prior to the Waste Load Allocation expiration date. An extension of the Waste Load Allocation will be for a period of up to an additional one (1) year and upon payment of an additional non-refundable waste load allocation fee, as well as other applicable fees that would be due based upon the allocated discharge limits as if the non-domestic user were discharging those limits even though the non-domestic user may not discharge until the operation permits are issued by SCDHEC pursuant to this ordinance. Within thirty (30) days from the date of Upon receipt of complete information, the Superintendent will determine whether or not to issue a wastewater discharge permit. The user may obtain the permit upon payment of the appropriate fees according to Sections 14 and 15 of this ordinance and the City of Cayce-Utility Rate Schedule.

5.2 Wastewater Discharge Permit Duration

A wastewater discharge permit shall be issued for a specified time period, not to exceed five (5) years from the effective date of the permit. A wastewater discharge permit may be issued for a period less than five (5) years, at the discretion of the Superintendent. Each wastewater discharge permit will indicate a specific date upon which it will expire.

5.3 Wastewater Discharge Permit Contents

A wastewater discharge permit shall include such conditions as are deemed reasonably necessary by the Superintendent to prevent pass through or interference, protect the quality of the water body receiving the treatment plant's effluent, protect worker health and safety, facilitate sludge management and disposal, and protect against damage to the POTW.

- 4.A. Wastewater discharge permits must contain:
 - →(1)A statement that indicates wastewater discharge permit duration, which in no event shall exceed five (5) years;
 - 2.(2) A statement that the wastewater discharge permit is nontransferable without prior notification

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

to the City in accordance with Section 5.6 of this ordinance, and provisions for furnishing the new owner or operator with a copy of the existing wastewater discharge permit;

- 3.(3) Effluent limits based on applicable pretreatment standards;
- 4-(4) Self monitoring, sampling, reporting, notification, and record-keeping requirements. These requirements shall include an identification of pollutants to be monitored, sampling location, sampling frequency, and sample type based on Federal, State, and local law; and
- 5-(5) A statement of applicable civil and criminal penalties for violation of pretreatment standards and requirements, and any applicable compliance schedule. Such schedule may not extend the time for compliance beyond that required by applicable Federal, State, or local law.
- 6-(6) Requirements to control Slug Discharge, if determined by the Superintendent to be necessary.
- 2.B. Wastewater discharge permits may contain the following conditions:
 - +(1) Limits on the average and/or maximum rate of discharge, time of discharge, and/or requirements for flow regulation and equalization;
 - 2-(2) Requirements for the installation of pretreatment technology, pollution control, or construction of appropriate containment devices, designed to reduce, eliminate, or prevent the introduction of pollutants into the treatment works;
 - 3-(3) Requirements for the development and implementation of spill control plans or other special conditions including management practices necessary to adequately prevent accidental, unanticipated, or nonroutine discharges;
 - 4-(4) Development and implementation of waste minimization plans to reduce the amount of pollutants discharged to the POTW;
 - 5-(5) The unit charge or schedule of user charges and fees for the management of the wastewater discharged to the POTW;
 - 6-(6) Requirements for installation and maintenance of inspection and sampling facilities and equipment;
 - 7-(7) A statement that compliance with the wastewater discharge permit does not relieve the permittee of responsibility for compliance with all applicable Federal and State pretreatment standards, including those which become effective during the term of the wastewater discharge permit; and
 - 8.(8) Other conditions as deemed appropriate by the Superintendent to ensure compliance with this ordinance, and State and Federal laws, rules, and regulations.

5.4 Wastewater Discharge Permit Appeals

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered +

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted

The Superintendent shall provide public notice of the issuance of a wastewater discharge permit. Any person, including the user, may petition the Superintendent to reconsider the terms of a wastewater discharge permit within fifteen (15) days of notice of its issuance.

- 4-A. Failure to submit a timely petition for review shall be deemed to be a waiver of the administrative appeal.
- 2-B. In its petition, the appealing party must indicate the wastewater discharge permit provisions objected to, the reasons for this objection, and the alternative condition, if any, it seeks to place in the wastewater discharge permit.
- 3-C. The effectiveness of the wastewater discharge permit shall not be stayed pending the appeal.
- 4-D. If the Superintendent fails to act within thirty (30) days, a request for reconsideration shall be deemed to be denied. Decisions not to reconsider a wastewater discharge permit, not to issue a wastewater discharge permit, or not to modify a wastewater discharge permit shall be considered final administrative actions.

5.5 Wastewater Discharge Permit Modification

The Superintendent may modify a wastewater discharge permit for good cause, including, but not limited to, the following reasons:

- A. To incorporate any new or revised Federal, State, or local pretreatment standards or requirements;
- 4-B. To address significant alterations or additions to the user's operation, processes, or wastewater volume or character since the time of wastewater discharge permit issuance;
- 2.C. A change in the POTW that requires either a temporary or permanent reduction or elimination of the authorized discharge;
- 3-D. Information indicating that the permitted discharge poses a threat to the City's POTW, City personnel, or the receiving waters;
- 4.E. Violation of any terms or conditions of the wastewater discharge permit;
- 5.F. Misrepresentations or failure to fully disclose all relevant facts in the wastewater discharge permit application or in any required reporting;
- 6-G. Revision of or a grant of variance from categorical pretreatment standards pursuant to 40 CFR 403.13;
- 7.H. To correct typographical or other errors in the wastewater discharge permit; or
- &.I. To reflect a transfer of the facility ownership or operation to a new owner or operator.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted

26

5.6 Wastewater Discharge Permit Transfer

Wastewater discharge permits may be transferred to a new owner or operator only if the permittee gives at least sixty (60) days advance notice to the Superintendent and the Superintendent approves the wastewater discharge permit transfer. The notice to the Superintendent must include a written certification by the new owner or operator which:

- 4-A. States that the new owner and/or operator has no immediate intent to change the facility's operations and processes;
- 2.B. Identifies the specific date on which the transfer is to occur; and
- 3-C. Acknowledges full responsibility for complying with the existing wastewater discharge permit.

Failure to provide advance notice of a transfer renders the wastewater discharge permit void as of the date of facility transfer.

5.7 Wastewater Discharge Permit Revocation

The Superintendent may revoke a wastewater discharge permit for good cause, including, but not limited to, the following reasons:

- 4-A. Failure to notify the Superintendent of significant changes to the wastewater prior to the changed discharge;
- 2-B. Failure to provide prior notification to the Superintendent of changed conditions pursuant to Section 6.5 of this ordinance;
- 3-C. Misrepresentation or failure to fully disclose all relevant facts in the wastewater discharge permit application;
- 4.D. Falsifying self-monitoring reports;
- 5.E. Tampering with monitoring equipment;
- 6.F. Refusing to allow the Superintendent timely access to the facility premises and records;
- 7.G. Failure to meet effluent limitations;
- 8.H. Failure to pay fines;
- 9-I. Failure to pay sewer charges;

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

27

- 10.J. Failure to comply with the Administrative Order or compliance schedules;
- 11.K. Failure to complete a wastewater survey or the wastewater discharge permit application;
- Failure to provide advance notice of the transfer of business ownership of a permitted facility;
- 13.M. Unauthorized dilution of wastewater in lieu of treatment; or
- 144.N. Violation of any pretreatment standard or requirement, or any terms of the wastewater discharge permit or this ordinance.

Wastewater discharge permits shall be voidable upon cessation of operations or transfer of business ownership. All wastewater discharge permits issued to a particular user are void upon the issuance of a new wastewater discharge permit to that user.

5.8 Wastewater Discharge Permit Reissuance

A user with an expiring wastewater discharge permit shall apply for wastewater discharge permit reissuance by submitting a complete permit application, in accordance with Section 4.5 of this ordinance, a minimum of six months (180 days) prior to the expiration of the user's existing wastewater discharge permit.

5.9 Regulation of Waste Received from Other Jurisdictions

- 4-A. If another municipality, or user located within another municipality, contributes wastewater to the POTW, the Superintendent shall enter into an intermunicipal agreement with the contributing municipality.
- 2-B. Prior to entering into an agreement required by paragraph A, above, the Superintendent shall request the following information from the contributing municipality:
 - 4-(1) A description of the quality and volume of wastewater discharged to the POTW by the contributing municipality;
 - 2...(2) An inventory of all users located within the contributing municipality that are discharging to the POTW; and
 - 3.(3) Such other information as the Superintendent may deem necessary.
- 3-C. An intermunicipal agreement, as required by paragraph A, above, shall contain the following conditions:
 - +(1)A requirement for the contributing municipality to adopt a sewer use ordinance which is at

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

least as stringent as this ordinance and local limits which are at least as stringent as those set out in Section 2.4 of this ordinance. The requirement shall specify that such ordinance and limits must be revised as necessary to reflect changes made to the City's ordinance or local limits;

- 2-(2) A requirement for the contributing municipality to submit a revised user inventory on at least an annual basis;
- 3-(3) A provision specifying which pretreatment implementation activities, including wastewater discharge permit issuance, inspection and sampling, and enforcement, will be conducted by the contributing municipality; which of these activities will be conducted by the Superintendent; and which of these activities will be conducted jointly by the contributing municipality and the Superintendent;
- 4-(4) A requirement for the contributing municipality to provide the Superintendent with access to all information that the contributing municipality obtains as part of its pretreatment activities:
- 5-(5) Limits on the nature, quality, and volume of the contributing municipality's wastewater at the point where it discharges to the POTW;
- 6.(6) Requirements for monitoring the contributing municipality's discharge;
- 7-(7) A provision ensuring the Superintendent access to the facilities of users located within the contributing municipality's jurisdictional boundaries for the purpose of inspection, sampling, and any other duties deemed necessary by the Superintendent;
- 8-(8) A provision specifying remedies available for breach of the terms of the intermunicipal agreement;

SECTION 6 - REPORTING REQUIREMENTS

6.1 Baseline Monitoring Reports

- 4.A. Within either one hundred eighty (180) days after the effective date of a categorical pretreatment standard, or the final administrative decision on a category determination under 40 CFR 403.6(a)(4), whichever is later, existing categorical users currently discharging to or scheduled to discharge to the POTW shall submit to the Superintendent a report which contains the information listed in paragraph B, below. At least ninety (90) days prior to commencement of their discharge, new sources, and sources that become categorical users subsequent to the promulgation of an applicable categorical standard, shall submit to the Superintendent a report which contains the information listed in paragraph B, below. A new source shall report the method of pretreatment it intends to use to meet applicable categorical standards. A new source also shall give estimates of its anticipated flow and quantity of pollutants to be discharged.
- 2.B. Categorical users shall submit the information set forth below.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

- 4-(1) <u>Identifying Information</u>. The name and address of the facility, including the name of the operator and owner.
- 2-(2) Environmental Permits. A list of any environmental control permits held by or for the facility.
- 3-(3) <u>Description of Operations</u>. A brief description of the nature, average rate of production, and standard industrial classifications of the operation(s) carried out by such user. This description should include a schematic process diagram which indicates points of discharge to the POTW from the regulated processes.
- 4-(4) Flow Measurement. Information showing the measured average daily and maximum daily flow, in gallons per day, to the POTW from regulated process streams and other streams, as necessary, to allow use of the combined wastestream formula set out in 40 CFR 403.6(e).
- 5.(5) Measurement of Pollutants.
 - 1.(a) The categorical pretreatment standards applicable to each regulated process.
 - 2-(b) The results of sampling and analysis identifying the nature and concentration, and/or mass, where required by the standard or by the Superintendent, of regulated pollutants in the discharge from each regulated process. Instantaneous, daily maximum, and long-term average concentrations, or mass, where required, shall be reported. The sample shall be representative of daily operations and shall be analyzed in accordance with procedures set out in Section 6.10 of this ordinance.
 - 3-(c) Sampling must be performed in accordance with procedures set out in Section 6.11 of this ordinance.
 - 4-(d) The user shall take a minimum of one representative sample to compile the data necessary to comply with the requirements of this section.
 - Samples should be taken immediately downstream from pretreatment facilities if such exist or immediately downstream from the regulated process if no pretreatment exists. If other wastewaters are mixed with the regulated wastewater prior to pretreatment the user should measure the flows and concentrations necessary to allow use of the combined waste stream formula in accordance with 40 CFR 403.6(e). Where process effluent is mixed prior to treatment with wastewaters other than those generated by the regulated process, fixed alternative discharge limits may be derived by the POTW or by the Industrial User with the written concurrence of the POTW. These alternative limits shall be applied to the mixed effluent. When deriving alternative categorical limits, the POTW or Industrial User shall calculate both an alternative daily maximum value using the daily maximum value(s) specified in the appropriate categorical Pretreatment Standard(s) and an alternative consecutive sampling day average value using the monthly average value(s) specified in the appropriate categorical Pretreatment Standard(s). The Industrial User shall comply with the alternative daily maximum and monthly average limits fixed by the POTW until the POTW modifies the limits or approves an Industrial User modification request. Modification is authorized whenever there is a material or significant change in the values used in the calculation to fix alternative limits for the

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15", Tab stops: Not at 1.5"

regulated pollutant. An Industrial User must immediately report any such material or significant change to the POTW where appropriate new alternative categorical limits shall be calculated within 30 days in order to evaluate compliance with Pretreatment Standards. Where an alternate concentration or mass limit has been calculated in accordance with 40 CFR 403.6(e) this adjusted limit along with supporting data shall be submitted to the POTW.

- 6-(6) Certification. A statement, reviewed by the user's authorized representative and certified by a qualified professional, indicating whether pretreatment standards are being met on a consistent basis, and, if not, whether additional operation and maintenance (O&M) and/or additional pretreatment is required to meet the pretreatment standards and requirements.
- 7-(7) Compliance Schedule. If additional pretreatment and/or O&M will be required to meet the pretreatment standards, the shortest schedule by which the user will provide such additional pretreatment and/or O&M. The completion date in this schedule shall not be later than the compliance date established for the applicable pretreatment standard. A compliance schedule pursuant to this section must meet the requirements set out in Section 6.2 of this ordinance.
- 8...(8) Signature and Certification. All baseline monitoring reports must be signed and certified in accordance with Section 4.6 of this ordinance.

6.2 Compliance Schedule Progress Reports

The following conditions shall apply to the compliance schedule required by Section 6.1 (B) (7) of this ordinance:

- 1-A. The schedule shall contain progress increments in the form of dates for the commencement and completion of major events leading to the construction and operation of additional pretreatment required for the user to meet the applicable pretreatment standards (such events include, but are not limited to, hiring an engineer, completing preliminary and final plans, executing contracts for major components, commencing and completing construction, and beginning and conducting routine operation);
- 2.B. No increment referred to above shall exceed nine (9) months;
- 3.C. The user shall submit a progress report to the Superintendent no later than fourteen (14) days following each date in the schedule and the final date of compliance including, as a minimum, whether or not it complied with the increment of progress, the reason for any delay, and, if appropriate, the steps being taken by the user to return to the established schedule; and
- 4-D. In no event shall more than nine (9) months elapse between such progress reports to the Superintendent

6.3 Reports on Compliance with Categorical Pretreatment Standard Deadline

Within ninety (90) days following the date for final compliance with applicable categorical pretreatment standards, or in the case of a new source following commencement of the introduction of

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9", Tab stops: Not at 1"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

wastewater into the POTW, any user subject to such pretreatment standards and requirements shall submit to the Superintendent a report containing the information described in Section 6.1(B)(4-6) of this ordinance. For users subject to equivalent mass or concentration limits established in accordance with the procedures in 40 CFR 403.6(c), this report shall contain a reasonable measure of the user's long-term production rate. For all other users subject to categorical pretreatment standards expressed in terms of allowable pollutant discharge per unit of production (or other measure of operation), this report shall include the user's actual production during the appropriate sampling period. All compliance reports must be signed and certified in accordance with Section 4.6 of this ordinance.

6.4 Periodic Compliance Reports

- 4-A. All significant industrial users shall, at a frequency determined by the Superintendent but in no case less than twice per year, submit a report indicating the nature and concentration of pollutants in the discharge which are limited by pretreatment standards and the measured or estimated average and maximum daily flows for the reporting period. In cases where the Pretreatment Standard requires compliance with a Best Management Practice or pollution prevention alternative, the User must submit documentation required by the Superintendent or the Pretreatment Standard necessary to determine the compliance status of the User. All periodic compliance reports must be signed and certified in accordance with Section 4.6 of this ordinance.
- 2-B. All wastewater samples must be representative of the user's discharge. Wastewater monitoring and flow measurement facilities shall be properly operated, kept clean, and maintained in good working order at all times. The failure of a user to keep its monitoring facility in good working order shall not be grounds for the user to claim that sample results are unrepresentative of its discharge.
- 3-C. If a user subject to the reporting requirement in this section monitors any pollutant more frequently than required by the Superintendent, using the procedures prescribed in Section 6.11 of this ordinance, the results of this monitoring shall be included in the report.

6.5 Reports of Changed Conditions

Each user must notify the Superintendent of any planned significant changes to the user's operations or system which might alter the nature, quality, or volume of its wastewater at least ninety (90) days before the change.

- +A. The Superintendent may require the user to submit such information as may be deemed necessary to evaluate the changed condition, including the submission of a wastewater discharge permit application under Section 4.5 of this ordinance.
- 2-B. The Superintendent may issue a wastewater discharge permit under Section 4.7 of this ordinance or modify an existing wastewater discharge permit under Section 5.5 of this ordinance in response to changed conditions or anticipated changed conditions.
- 3.C. For purposes of this requirement, significant changes include, but are not limited to, flow increases

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

of twenty percent (20%) or greater, and the discharge of any previously unreported pollutants.

6.6 Reports of Potential Problems

- 1-A. In the case of any discharge, including, but not limited to, accidental discharges, discharges of a nonroutine, episodic nature, a noncustomary batch discharge, or a slug load, that may cause potential problems for the POTW, the user shall immediately notify the Superintendent of the incident by telephone or facsimile or by email. This notification shall include the location of the discharge, type of waste, concentration and volume, if known, and corrective actions taken by the user
- 2-B. Within five (5) days following such discharge, the user shall, unless waived by the Superintendent, submit a detailed written report describing the cause(s) of the discharge and the measures to be taken by the user to prevent similar future occurrences. Such notification shall not relieve the user of any expense, loss, damage, or other liability which may be incurred as a result of damage to the POTW, natural resources, or any other damage to person or property; nor shall such notification relieve the user of any fines, penalties, or other liability which may be imposed pursuant to this ordinance.
- 3.C. A notice shall be permanently posted on the user's bulletin board or other prominent place advising employees whom to call in the event of a discharge described in paragraph A above. Employers shall ensure that all employees who may cause such a discharge to occur are advised of the emergency notification procedure.

6.7 Reports from Unpermitted Users

All users not required to obtain a wastewater discharge permit shall provide appropriate reports to the Superintendent as the Superintendent may require.

6.8 Notice of Violation/Repeat Sampling and Reporting

If sampling performed by a user indicates a violation, the user must notify the Superintendent in writing within twenty-four (24) hours of becoming aware of the violation. The user shall also repeat the sampling and analysis and submit the results of the repeat analysis to the Superintendent within thirty (30) days after becoming aware of the violation. The user is not required to resample if the Superintendent monitors at the user's facility at least once a month, or if the Superintendent samples between the user's initial sampling and when the user receives the results of this sampling. If the City performed the sampling and analysis in lieu of the User, the City will perform the repeat sampling and analysis unless it notifies the User of the violation and requires the User to perform the repeat sampling and analysis. Failure to notify the Superintendent of a violation will result in enforcement action being taken in accordance with the City's approved Enforcement Response Plan.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

6.9 Notification of the Discharge of Hazardous Waste

- +A. Any user who commences the discharge of hazardous waste shall notify the Superintendent, the EPA Regional Waste Management Division Director, and State hazardous waste authorities, in writing, of any discharge into the POTW of a substance which, if otherwise disposed of, would be a hazardous waste under 40 CFR Part 261. Such notification must include the name of the hazardous waste as set forth in 40 CFR Part 261, the EPA hazardous waste number, and the type of discharge (continuous, batch, or other). If the user discharges more than one hundred (100) kilograms of such waste per calendar month to the POTW, the notification also shall contain the following information to the extent such information is known and readily available to the user: an identification of the hazardous constituents contained in the wastes, an estimation of the mass and concentration of such constituents in the wastestream discharged during that calendar month, and an estimation of the mass of constituents in the wastestream expected to be discharged during the following twelve (12) months. All notifications must take place no later than thirty (30) days after the discharge commences. Any notification under this paragraph need be submitted only once for each hazardous waste discharged. However, notifications of changed conditions must be submitted under Section 6.5 of this ordinance. The notification requirement in this section does not apply to pollutants already reported by users subject to categorical pretreatment standards under the self-monitoring requirements of Sections 6.1, 6.3, and 6.4 of this ordinance.
- 2-B. Dischargers are exempt from the requirements of paragraph A, above, during a calendar monthin which they discharge no more than fifteen (15) kilograms of hazardous wastes, unless the wastes are acute hazardous wastes as specified in 40 CFR 261.30(d) and 261.33(e). Discharge of more than fifteen (15) kilograms of nonacute hazardous wastes in a calendar month, or of any quantity of acute hazardous wastes as specified in 40 CFR 261.30(d) and 261.33(e), requires a one-time notification. Subsequent months during which the user discharges more than such quantities of any hazardous waste do not require additional notification.
- 3.C. In the case of any new regulations under Section 3001 of RCRA identifying additional characteristics of hazardous waste or listing any additional substance as a hazardous waste, the user must notify the Superintendent, the EPA Regional Waste Management Waste Division Director, and State hazardous waste authorities of the discharge of such substance within ninety (90) days of the effective date of such regulations.
- 4-D. In the case of any notification made under this section, the user shall certify that it has a program in place to reduce the volume and toxicity of hazardous wastes generated to the degree it has determined to be economically practical.
- 5-E. This provision does not create a right to discharge any substance not otherwise permitted to be discharged by this ordinance, a permit issued thereunder, or any applicable Federal or State law.

6.10 Analytical Requirements

All pollutant analyses, including sampling techniques, to be submitted as part of a wastewater discharge permit application or report shall be performed in accordance with the techniques prescribed in 40 CFR Part 136, unless otherwise specified in an applicable categorical pretreatment standard. If 40 CFR Part

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

136 does not contain sampling or analytical techniques for the pollutant in question, sampling and analyses must be performed in accordance with procedures approved by SCDHEC. The user shall report, to the City of all the sample analyses conducted according to 40 CFR Part 136 or SCDHEC approved procedures, including those that are not designed for periodic compliance reporting under Section 6.4 of this ordinance.

6.11 Sample Collection

- +A. Except as indicated in Section B, below, the user must collect wastewater samples using 24-hour flow proportional composite samples. In the event flow proportional sampling is infeasible, the Superintendent may authorize the use of time proportional sampling or a minimum of four (4) grab samples where the user demonstrates that this will provide a representative sample of the effluent being discharged. In addition, grab samples may be required to show compliance with instantaneous discharge limits.
- 2-B. Samples for oil and grease, temperature, pH, cyanide, total phenols, sulfides, and volatile organic compounds must be obtained using grab collection techniques.
- 3-C. For sampling required in support of baseline monitoring and 90-day compliance reports required in Section 6.1 and 6.3, a minimum of four (4) grab samples must be used for pH, cyanide, total phenols, oil and grease, sulfide, and volatile organic compounds for facilities for which historical sampling data do not exist; for facilities for which historical sampling data are available, the Superintendent may authorize a lower minimum. For the reports required by Section 6.4, the User is required to collect the number of grab samples necessary to assess and assure compliance with applicable Pretreatment Standards and Requirements.

6.12 Timing

Written reports will be deemed to have been submitted on the date postmarked. For reports which are not mailed, postage prepaid, into a mail facility serviced by the United States Postal Service, the date of receipt of the report shall govern.

6.13 Record Keeping

Users subject to the reporting requirements of this ordinance shall retain, and make available for inspection and copying, all records of information obtained pursuant to any monitoring activities required by this ordinance and any additional records of information obtained pursuant to monitoring activities undertaken by the user independent of such requirements and documentation associated with Best Management Practices. Records shall include the date, exact place, method, and time of sampling, and the name of the person(s) taking the samples; the dates analyses were performed; who performed the analyses; the analytical techniques or methods used; and the results of such analyses. These records shall remain available for a period of at least three (3) years. This period shall be automatically extended for the duration of any litigation concerning the user or the City, or where the user has been specifically notified of a longer retention period by the Superintendent.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

SECTION 7 - COMPLIANCE MONITORING

7.1 Right of Entry: Inspection and Sampling

The Superintendent shall have the right to enter the premises of any user to determine whether the user is complying with all requirements of this ordinance and any wastewater discharge permit or order issued hereunder. Users shall allow the Superintendent ready access to all parts of the premises for the purposes of inspection, sampling, records examination and copying, and the performance of any additional duties.

- 4-A. Where a user has security measures in force which require proper identification and clearance before entry into its premises, the user shall make necessary arrangements with its security guards so that, upon presentation of suitable identification, the Superintendent will be permitted to enter without delay for the purposes of performing specific responsibilities.
- 2-B. The Superintendent shall have the right to set up on the user's property, or require installation of, such devices as are necessary to conduct sampling and/or metering of the user's operations.
- 3.C. The Superintendent may require the user to install monitoring equipment as necessary. The facility's sampling and monitoring equipment shall be maintained at all times in a safe and proper operating condition by the user at its own expense. All flow measuring devices shall be calibrated at least annually or according to the manufacturer's recommendations, whichever is more stringentonce every six months. To insure accuracy, all water quality analysis shall be justified using blanks.
- 4-D. Any temporary or permanent obstruction to safe and easy access to the facility to be inspected and/or sampled shall be promptly removed by the user at the written or verbal request of the Superintendent and shall not be replaced. The costs of clearing such access shall be born by the user.
- 5-E. Unreasonable delays in allowing the Superintendent access to the user's premises shall be a violation of this ordinance.

7.2 Search Warrants

If the Superintendent has been refused access to a building, structure, or property, or any part thereof, and is able to demonstrate probable cause to believe that there may be a violation of this ordinance, or that there is a need to inspect and/or sample as part of a routine inspection and sampling program of the City designed to verify compliance with this ordinance or any permit or order issued hereunder, or to protect the overall public health, safety and welfare of the community, then the Superintendent may seek issuance of a search warrant.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59", Tab stops: Not at 0.5"

SECTION 8 - CONFIDENTIAL INFORMATION

Information and data on a user obtained from reports, surveys, wastewater discharge permit applications, wastewater discharge permits, and monitoring programs, and from the Superintendent's inspection and sampling activities, shall be available to the public without restriction, unless the user specifically requests, and is able to demonstrate to the satisfaction of the Superintendent, that the release of such information would divulge information, processes, or methods of production entitled to protection as trade secrets under applicable State law. Any such request must be asserted at the time of submission of the information or data. When requested and demonstrated by the user furnishing a report that such information should be held confidential, the portions of a report which might disclose trade secrets or secret processes shall not be made available for inspection by the public, but shall be made available immediately upon request to governmental agencies for uses related to the NPDES program or pretreatment program, and in enforcement proceedings involving the person furnishing the report. Wastewater constituents and characteristics and other "effluent data" as defined by 40 CFR 2.302 will not be recognized as confidential information and will be available to the public without restriction.

SECTION 9 - PUBLICATION OF USERS IN SIGNIFICANT NONCOMPLIANCE

The Superintendent shall publish annually, in a newspaper of general circulation in the municipality where the POTW is located, a list of the Users which, at any time during the previous twelve (12) months, were in Significant Noncompliance with applicable Pretreatment Standards and Requirements. The term Significant Noncompliance shall be applicable to all Significant Industrial Users (or any other User that violates paragraphs (C), (D), or (H) of this Section) and shall mean:

- A. Chronic violations of wastewater discharge limits, defined here as those in which sixty-six percent (66%) or more of all the measurements taken for the same pollutant parameter during a six- (6-) month period exceed (by any magnitude) a numeric Pretreatment Standard or Requirement, including instantaneous limits, as defined by 40 CFR 403.3(1);
- B. Technical Review Criteria (TRC) violations, defined here as those in which thirty-three percent (33%) or more of all of the measurements taken for the same pollutant parameter during a six- (6-) month period equals or exceeds the product of the numeric Pretreatment Standard or Requirement including instantaneous limits as defined by 40 CFR 403.3(1) multiplied by the applicable TRC (TRC=1.4 for BOD, TSS, fats, oils and grease, and 1.2 for all other pollutants except pH);
- C. Any other violation of a Pretreatment Standard or Requirement as defined by 40 CFR 403.3(1) (daily maximum, long term average, instantaneous limit, or narrative Standard) that the POTW determines has caused, alone or in combination with other discharges, interference or pass through,

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

including endangering the health of POTW personnel or the general public;

- Any discharge of a pollutants that has caused imminent endangerment to human health, welfare
 or to the environment, or has resulted in the POTW's exercise of its emergency authority to halt
 or prevent such a discharge;
- E. Failure to meet, within ninety (90) days after of the scheduled date, a compliance schedule milestone contained in a local control mechanism or enforcement order for starting construction, completing construction, or attaining final compliance;
- F. Failure to provide within thirty (30) days after the due date, any required reports, such as baseline monitoring reports, ninety (90) day compliance reports, periodic self-monitoring reports, and reports on compliance with compliance schedules;
- G. Failure to accurately report noncompliance; or
- H. Any other violation(s), which may include a violation of the Best Management Practices, which the POTW determines will adversely affect the operation or implementation of the local pretreatment program.

SECTION 10 - ADMINISTRATIVE ENFORCEMENT REMEDIES

10.1 Notification of Violation

When the Superintendent finds that a user has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Superintendent may serve upon that user a written Notice of Violation. Within ten (10) days of the receipt of this notice, an explanation of the violation and a plan for the satisfactory correction and prevention thereof, to include specific required actions, shall be submitted by the user to the Superintendent. Submission of this plan in no way relieves the user of liability for any violations occurring before or after receipt of the Notice of Violation. Nothing in this section shall limit the authority of the Superintendent to take any action, including emergency actions or any other enforcement action, without first issuing a Notice of Violation.

10.2 Consent Orders

The Superintendent may enter into Consent Orders, assurances of voluntary compliance, or other similar documents establishing an agreement with any user responsible for noncompliance. Such documents will include specific action to be taken by the user to correct the noncompliance within a time period specified by the document. Such documents shall have the same force and effect as the administrative orders issued pursuant to Sections 10.4 and 10.5 of this ordinance and shall be judicially enforceable.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

10.3 Show Cause Hearing

The Superintendent may order a user which has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, to appear before the Superintendent and show cause why the proposed enforcement action should not be taken. Notice shall be served on the user specifying the time and place for the meeting, the proposed enforcement action, the reasons for such action, and a request that the user show cause why the proposed enforcement action should not be taken. The notice of the meeting shall be served personally or by registered or certified mail (return receipt requested) at least seven (7) days prior to the hearing. Such notice may be served on any authorized representative of the user. A show cause hearing shall not be a bar against, or prerequisite for, taking any other action against the user.

10.4 Compliance Orders

When the Superintendent finds that a user has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Superintendent may issue an order to the user responsible for the discharge directing that the user come into compliance within a specified time. If the user does not come into compliance within the time provided, sewer service may be discontinued unless adequate treatment facilities, devices, or other related appurtenances are installed and properly operated. Compliance orders also may contain other requirements to address the noncompliance, including additional self-monitoring and management practices designed to minimize the amount of pollutants discharged to the sewer. A compliance order may not extend the deadline for compliance established for a pretreatment standard or requirement, nor does a compliance order relieve the user of liability for any violation, including any continuing violation. Issuance of a compliance order shall not be a bar against, or a prerequisite for, taking any other action against the user.

10.5 Cease and Desist Orders

When the Superintendent finds that a user has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, or that the user's past violations are likely to recur, the Superintendent may issue an order to the user directing it to cease and desist all such violations and directing the user to:

(1)A. ___Immediately comply with all requirements; and

2)B. Take such appropriate remedial or preventive action as may be needed to properly address

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

a continuing or threatened violation, including halting operations and/or terminating the discharge.

Issuance of a cease and desist order shall not be a bar against, or a prerequisite for, taking any other action against the user.

10.6 Administrative Fines

- A. When the Superintendent finds that a user has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit or order issued hereunder, or any other pretreatment standard or requirement, the Superintendent may fine such user in an amount not less than one hundred dollars (\$100.00) and not to exceed two-thousand dollars (\$2000.00). Such fines shall be assessed on a per violation, per day basis. In the case of a continuing violation, each day's offense shall be counted as a separate and distinct offense. Fines for a continuing violation shall be assessed for each day during the period of violation. Depending on the severity of the violation, the City may elect to take other enforcement action(s) with or without imposing a fine. In the case of recurring violations of a similar nature, escalation from a Notice of Violation to an administrative fine will occur after three (3) occurrences in a consecutive six (6) month period.
- B. Unpaid charges, fines, and penalties shall, after thirty (30) calendar days, be assessed an additional penalty of five percent (5%) of the unpaid balance, and interest shall accrue thereafter at a rate of one percent (1%) per month. A lien against the user's property will be sought for unpaid charges, fines, and penalties.
- C. Users desiring to dispute such fines must file a written request for the Superintendent to reconsider the fine along with full payment of the fine amount within thirty (30) days of being notified of the fine. Where a request has merit, the Superintendent may convene a hearing on the matter. In the event the user's appeal is successful, the payment, together with any interest accruing thereto, shall be returned to the user. The Superintendent may add the costs of preparing administrative enforcement actions, such as notices and orders, to the fine.
- D. Issuance of an administrative fine shall not be a bar against, or a prerequisite for, taking any other action against the user.

10.7 Emergency Suspensions

The Superintendent may immediately suspend a user's discharge, after informal notice to the user, whenever such suspension is necessary to stop an actual or threatened discharge which reasonably appears to present or cause an imminent or substantial endangerment to the health or welfare of persons. The Superintendent may also immediately suspend a user's discharge, after notice and opportunity to respond, that threatens to interfere with the operation of the POTW, or which presents, or may present, an endangerment to the environment.

A. Any user notified of a suspension of its discharge shall immediately stop or eliminate its

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

contribution. In the event of a user's failure to immediately comply voluntarily with the suspension order, the Superintendent may take such steps as deemed necessary, including immediate severance of the sewer connection, to prevent or minimize damage to the POTW, its receiving stream, or endangerment to any individuals. The Superintendent may allow the user to recommence its discharge when the user has demonstrated to the satisfaction of the Superintendent that the period of endangerment has passed, unless the termination proceedings in Section 10.8 of this ordinance are initiated against the user.

B. A user that is responsible, in whole or in part, for any discharge presenting imminent endangerment shall submit a detailed written statement, describing the causes of the harmful contribution and the measures taken to prevent any future occurrence, to the Superintendent prior to the date of any show cause or termination hearing under Sections 10.3 or 10.8 of this ordinance.

Nothing in this section shall be interpreted as requiring a hearing prior to any emergency suspension under this section.

10.8 Termination of Discharge

In addition to the provisions in Section 5.7 of this ordinance, any user who violates the following conditions is subject to discharge termination:

- A. Violation of wastewater discharge permit conditions;
- B. Failure to accurately report the wastewater constituents and characteristics of its discharge;
- Failure to report significant changes in operations or wastewater volume, constituents, and characteristics prior to discharge;
- Refusal of reasonable access to the user's premises for the purpose of inspection, monitoring, or sampling;
- E. Violation of the pretreatment standards in Section 2 of this ordinance;
- F. Falsifying reports;
- G. Failure to install monitoring equipment (intentional);
- H. Improper sampling scheme;
- Violation of compliance schedule; or
- J. Wastewater dilution, in lieu of treatment.

Such user will be notified of the proposed termination of its discharge and be offered an opportunity to show cause under Section 10.3 of this ordinance why the proposed action should not be taken. Exercise of this option by the Superintendent shall not be a bar to, or a prerequisite for, taking any other action

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

SECTION 11 - JUDICIAL ENFORCEMENT REMEDIES

11.1 Injunctive Relief

When the Superintendent finds that a user has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, the Superintendent may petition Lexington County Court of Common Pleas through the City's Attorney for the issuance of a temporary or permanent injunction, as appropriate, which restrains or compels the specific performance of the wastewater discharge permit, order, or other requirement imposed by this ordinance on activities of the user. The Superintendent may also seek such other action as is appropriate for legal and/or equitable relief, including a requirement for the user to conduct environmental remediation. A petition for injunctive relief shall not be a bar against, or a prerequisite for, taking any other action against a user.

11.2 Civil Penalties

- 4)A. A user who has violated, or continues to violate, any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall be liable to the City for a civil penalty of not more than two-thousand dollars (\$2000.00), pursuant to South Carolina Code Section 6-11-285, per violation per day. In the case of a monthly or other long-term average discharge limit, penalties shall accrue for each day during the period of the violation.
- The Superintendent may recover reasonable attorneys' fees, court costs, and other expenses associated with enforcement activities, including medical treatment of injured City employees, sampling and monitoring expenses, and the cost of any actual damages incurred by the City.
- In determining the amount of civil liability, the Court shall take into account all relevant circumstances, including, but not limited to, the extent of harm caused by the violation, the magnitude and duration of the violation, any economic benefit gained through the user's violation, corrective actions by the user, the compliance history of the user, and any other factor as justice requires.
- (4)D. Filing a suit for civil penalties shall not be a bar against, or a prerequisite for, taking any other action against a user.

11.3 Criminal Prosecution

A. A user who willfully or negligently violates any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement shall, upon conviction, be guilty of a misdemeanor, punishable by a fine of not more than two-thousand

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

dollars (\$2000.00) per violation, per day, or imprisonment for not more than two years, or both.

11.4 Remedies Nonexclusive

The remedies provided for in this ordinance are not exclusive. The Superintendent may take any, all, or any combination of these actions against a noncompliant user. Enforcement of pretreatment violations will generally be in accordance with the City's enforcement response plan. However, the Superintendent may take other action against any user when the circumstances warrant. Further, the Superintendent is empowered to take more than one enforcement action against any noncompliant user.

SECTION 12 - SUPPLEMENTAL ENFORCEMENT ACTION

12.1 Performance Bonds

The Superintendent may decline to issue or reissue a wastewater discharge permit to any user who has failed to comply with any provision of this ordinance, a previous wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, unless such user first files a satisfactory bond, payable to the City, in a sum not to exceed a value determined by the Superintendent to be necessary to achieve consistent compliance.

12.2 Liability Insurance

The Superintendent may decline to issue or reissue a wastewater discharge permit to any user who has failed to comply with any provision of this ordinance, a previous wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, unless the user first submits proof that it has obtained financial assurances sufficient to restore or repair damage to the POTW caused by its discharge.

12.3 Water Supply Severance

Whenever a user has violated or continues to violate any provision of this ordinance, a wastewater discharge permit, or order issued hereunder, or any other pretreatment standard or requirement, water service to the user may be severed. Service will only recommence, at the user's expense, after it has satisfactorily demonstrated its ability to comply.

12.4 Public Property Damage

A violation of any provision of this ordinance, a wastewater discharge permit, or order issued

hereunder, or any other pretreatment standard or requirement is hereby declared a damage to public property and shall be corrected or abated as directed by the Superintendent. Any person(s) creating damage to public property shall be subject to the provisions of the City of Cayce Code 28-42 Tampering with Property of Public Utilities and Code 28-54 Injuring Public Property and Property Open to Public, governing such damage, including reimbursing the City for any costs incurred in removing, abating, or remedying said damage.

12.5 Informant Rewards

The Superintendent may pay up to five hundred dollars (\$500.00) for information leading to the discovery of noncompliance by a user. In the event that the information provided results in a civil penalty or an administrative fine levied against the user, the Superintendent may disperse up to ten percent (10%) of the collected fine or penalty to the informant. However, a single reward payment may not exceed three thousand dollars (\$3,000).

12.6 Contractor Listing

Users which have not achieved compliance with applicable pretreatment standards and requirements are not eligible to receive a contractual award for the sale of goods or services to the City. Existing contracts for the sale of goods or services to the City held by a user found to be in significant noncompliance with pretreatment standards or requirements may be terminated at the discretion of the Superintendent.

SECTION 13 - AFFIRMATIVE DEFENSES TO DISCHARGE VIOLATIONS

13.1 Upset

- 1)A. For the purposes of this section, "upset" means an exceptional incident in which there is unintentional and temporary noncompliance with categorical pretreatment standards because of factors beyond the reasonable control of the user. An upset does not include noncompliance to the extent caused by operational error, improperly designed treatment facilities, inadequate treatment facilities, lack of preventive maintenance, or careless or improper operation.
- (2)B. An upset shall constitute an affirmative defense to an action brought for noncompliance with categorical pretreatment standards if the requirements of paragraph (C), below, are met.
- A user who wishes to establish the affirmative defense of upset shall demonstrate, through properly signed, contemporaneous operating logs, or other relevant evidence that:
 - 4.(1) An upset occurred and the user can identify the cause(s) of the upset;

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

- 2-(2) The facility was at the time being operated in a prudent and workman-like manner and in compliance with applicable operation and maintenance procedures; and
- 3-(3) The user has submitted the following information to the Superintendent within twenty-four (24) hours of becoming aware of the upset. If this information is provided orally, a written submission must be provided within five (5) days:
 - A.(a) A description of the indirect discharge and cause of noncompliance;
 - B.(b) The period of noncompliance, including exact dates and times or, if not corrected, the anticipated time the noncompliance is expected to continue; and
 - C:(c) Steps being taken and/or planned to reduce, eliminate, and prevent recurrence of the noncompliance.
- In any enforcement proceeding, the user seeking to establish the occurrence of an upset shall have the burden of proof.
- (S)E. Users will have the opportunity for a judicial determination on any claim of upset only in an enforcement action brought for noncompliance with categorical pretreatment standards.
- (6)F. Users shall control production of all discharges to the extent necessary to maintain compliance with categorical pretreatment standards upon reduction, loss, or failure of its treatment facility until the facility is restored or an alternative method of treatment is provided. This requirement applies in the situation where, among other things, the primary source of power of the treatment facility is reduced, lost, or fails.

13.2 Prohibited Discharge Standards

A user shall have an affirmative defense to an enforcement action brought against it for noncompliance with the general prohibitions in Section 2.2(A) of this ordinance or the specific prohibitions in Sections 2.2(B)(3) through 2.2(B)(6) of this ordinance if it can prove that it did not know, or have reason to know, that its discharge, alone or in conjunction with discharges from other sources, would cause pass through or interference and that either:

- A. A local limit exists for each pollutant discharged and the user was in compliance with each limit directly prior to, and during, the pass through or interference; or
- B. No local limit exists, but the discharge did not change substantially in nature or constituents from the user's prior discharge when the City was regularly in compliance with its NPDES permit, and in the case of interference, was in compliance with applicable sludge use or disposal requirements.

13.3 Bypass

A. For the purposes of this section,

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

- "Bypass" means the intentional diversion of wastestreams from any portion of a user's treatment facility.
- "Severe property damage" means substantial physical damage to property, damage to the treatment facilities which causes them to become inoperable, or substantial and permanent loss of natural resources which can reasonably be expected to occur in the absence of a bypass. Severe property damage does not mean economic loss caused by delays in production.
- B. A user may allow any bypass to occur which does not cause pretreatment standards or requirements to be violated, but only if it also is for essential maintenance to assure efficient operation. These bypasses are not subject to the provision of paragraphs (C) and (D) of this section.
- C. (1) If a user knows in advance of the need for a bypass, it shall submit prior notice to the Superintendent, at least ten (10) days before the date of the bypass, if possible.
 - (2) A user shall submit oral notice to the Superintendent of an unanticipated bypass that exceeds applicable pretreatment standards within twenty-four (24) hours from the time it becomes aware of the bypass. A written submission shall also be provided within five (5) days of the time the user becomes aware of the bypass. The written submission shall contain a description of the bypass and its cause; the duration of the bypass, including exact dates and times, and, if the bypass has not been corrected, the anticipated time it is expected to continue; and steps taken or planned to reduce, eliminate, and prevent reoccurrence of the bypass. The Superintendent may waive the written report on a case-by-case basis if the oral report has been received within twenty-four (24) hours.
- D. (1) Bypass is prohibited, and the Superintendent may take an enforcement action against a user for a bypass, unless
 - A-(a) Bypass was unavoidable to prevent loss of life, personal injury, or severe property damage;
 - There were no feasible alternatives to the bypass, such as the use of auxiliary treatment facilities, retention of untreated wastes, or maintenance during normal periods of equipment downtime. This condition is not satisfied if adequate back-up equipment should have been installed in the exercise of reasonable engineering judgment to prevent a bypass which occurred during normal periods of equipment downtime or preventive maintenance; and
 - C.(c) The user submitted notices as required under paragraph (C) of this section.
 - (2) The Superintendent may approve an anticipated bypass, after considering its adverse effects, if the Superintendent determines that it will meet the three conditions listed in paragraph (D)(1) of this section.

SECTION 14 - WASTEWATER TREATMENT RATES

To provide facilities for conveying and treating the wastewater, to recover the annual operating costs, and to provide sludge management costs, the City adopts the following types of fees. The amount of

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Left: 0.28", Hanging: 0.63", Numbered + Level: 1 + Numbering Style: A, B, C, ... - Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Left: 0.28", Hanging: 0.63", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 1 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

each of the fees established herein is detailed in the most recent City of Cayce-Utility Rate Schedule which is updated from time to time. All users shall pay the appropriate fee as determined by the Superintendent based upon the rates in the Utility Rate Schedule and the type of service being provided.

14.1 Fees

- •(1). Application Fee. A user shall pay a fee at the time of initial application for service.
- <u>a(2). Survey Questionnaire Fee.</u> Fee for the analysis of waste survey information for the proposed wastewater discharge to determine if the user must have a wastewater contribution permit.
- <u>a(3). Capacity Reservation Fee</u>. Fees for providing a reservation of capacity in the City POTW for a specific period of time.
- •(4). Tap Fee. Fees for providing the connection to the City POTW.
- <u>*(5). Impact Fee (Capacity Charge)</u>. Fees for the impact of wastewater discharges on the City POTW system. The impact on the City POTW will be assessed using the Guidelines for Unit Contributory Loadings as published by SCDHEC (latest edition).
- <u>a(6). Sewer Availability Fees.</u> City may charge for sewer when sewer service is available adjacent to any property in use, or otherwise occupied, even if a property is not discharging wastewater to the City sewer.
- <u>*(7). Monthly Usage Fees.</u> All users are subjected to a monthly fee based on the rate system adopted by the City.
- <u>a(8). Surcharge Fees.</u> All industrial users shall be subjected to the following types of surcharges. The exact amount is based on the current rates adopted by the City.
 - •(a)BOD exceeding 250 mg/l
 - (b)TSS exceeding 250 mg/l
 - (c)Other pollutants, if available, in amounts exceeding the "pro-rata" share based upon the flow using the allowable remaining balance as determined by the Headworks Analysis and making an equitable distribution of the unused balance of pollutants to all users requesting additional capacity and subject to the approval of the Superintendent.
- A.(9) Permit Fees. The City may charge FSEs and waste haulers a wastewater discharge fee for review of plans or other documentation, processing, and issuance of wastewater discharge permits associated with the disposal of septage and/or grease.
- B.(10) Dumping Fees. The City may charge a per gallon fee to Industrial, Septic and Grease Haulers

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 3 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 9 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 9 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

14.2 Measurement

A.(1) Monthly Usage Measurement.

- <u>a(a)</u>The City may install a separate meter for sewer service to the user for measurement of the monthly sewer flow.
- (b) If the user is also a City water user, and the City feels that it is impractical to install separate flow measurement equipment for the sewer line serving the user, it is permissible to use the water meter serving the user as the basis for the monthly sewer fee.
- •(c) In the event that the user is not a water customer or has no water meter, the City may at its discretion make a determination of a flat rate monthly fee based upon similar types of users and their discharges or may monitor flows temporarily to obtain an average flow as a basis for the monthly charge, install a meter in well water supply line at owners expense, or use any other equitable method of determining an average usage.
- •(d) If the City determines that it is impractical to install a separate meter for measurement of sewer flow and uses any alternate method for determination of monthly sewer fees, the user has the right to pay the City for all costs to install a separate meter for sewer. In addition to the costs for the installation of the meter there will be a separate monthly fee for maintenance and reading of the separate meter.
- B-(2) Surcharge Measurement. Surcharges shall be billed on a monthly basis to all permitted users who have additional capacity of any pollutant beyond the pro-rata share based on flows as determined by the headworks analysis. The monthly Surcharge Fee shall be based on the permitted amount of each pollutant or on the actual discharge of any pollutant, whichever is greater. However, if the amount of pollutant discharged is in excess of the permitted amount there may be fines and penalties in accordance with Section 10 of this Ordinance.

SECTION 15 - MISCELLANEOUS PROVISIONS

14.3 Pretreatment Charges and Fees

To provide for reimbursement of the development cost of the City's Pretreatment Program, for recovery of the annual operating costs, and for maintenance of the pretreatment program, the City herein adopts the following categories of fees. The amount of each of the fees established herein is detailed in the most recent City of Cayce-Utility Rate Schedule which is updated from time to time. All users shall pay the appropriate fees as determined by the Superintendent based upon the rates in the Utility Rate Schedule and the types of service being provided.

- A. <u>Pretreatment Application Fees</u>. Fees for wastewater discharge permit applications including the cost of processing such applications in the following two part process:
 - (1) Waste Load Allocation Fee. A non-refundable fee paid by the non-domestic user prior to the issuance of the Waste Load Allocation. The Waste Load Allocation insures the non-domestic user of the capacity and loading limits in the City POTW for a period of one (1) year, within which time final plans or sufficient documentation must be submitted to the City and a Wastewater Discharge Permit must be issued.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Numbered + Level: 2 + Numbering Style: a, b, c, ... + Start at: 1 + Alignment: Left + Aligned at: 0.9" + Indent at: 1.15"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

- (2) Wastewater Discharge Permit Fee. Fee for review of final plans or other documentation, processing, and issuance of pretreatment permits.
- B. Annual Fee. Fees for monitoring, inspection, and surveillance procedures including the cost of collection and analyzing a user's discharge, and reviewing monitoring reports submitted by users. This fee is billed at the beginning of each fiscal year, in the first week of the month of July. Failure to pay the required dues by the due date will result in fines and/or surcharges.
- C. Fees for reviewing and responding to accidental discharge procedures and construction.
- D. Fees for filing appeals.
- E. Other fees as the City may deem necessary to carry out the requirements contained herein. These fees relate solely to the matters covered by this ordinance and are separate from all other fees, fines, and penalties chargeable by the City.

14.4 Severability

If any provision of this ordinance is invalidated by any court of competent jurisdiction, the remaining provisions shall not be effected and shall continue in full force and effect.

SECTION 46-15 - EFFECTIVE DATE

This ordinance shall be in full force and effect immediately following its passage, approval, and publication, as provided by law. This ordinance shall supersede the previous sewer use ordinance and amendments thereof.

Formatted: Indent: Hanging: 0.31", Numbered + Level: 2 + Numbering Style: 1, 2, 3, ... + Start at: 1 + Alignment: Left + Aligned at: 0.59" + Indent at: 0.9"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

Formatted: Indent: Hanging: 0.31", Numbered + Level: 1 + Numbering Style: A, B, C, ... + Start at: 1 + Alignment: Left + Aligned at: 0.27" + Indent at: 0.59"

City of Cayce Fats, Oil and Grease Control Policy Section

1: Purpose and Objectives:

The objective of this Policy is to aid in administration of the City of Cayce's Fats, Oils, and Grease Ordinance ("FOG Ordinance") designed to reduce the introduction and accumulation of fats, oils, and greases into the municipal wastewater collection system.

This Policy establishes uniform construction, maintenance and monitoring requirements for controlling the discharge of grease from food service facilities discharging into the wastewater collection system and for regulation of commercial grease waste haulers operating within the City of Cayce's (City's) collection system.

Section 2: Definitions: The following definitions apply to this Policy.

<u>City</u> means the City of Cayce Utilities Department

<u>Fats, oils, and greases</u> means organic polar compounds derived from animal and/or plant sources that contain multiple carbon chain triglyceride molecules and non polar compounds (petroleum origin) which are in excess of one hundred (100) mg/l or containing substances which may solidify or become viscous at temperatures between thirty-two (32) and one hundred fifty (150) degrees Fahrenheit (zero (0) and sixty five (65) degrees centigrade). These substances are detectable and measurable using analytical test procedures established in 40 CFR 136, as may be amended from time to time. All are sometimes referred to herein as "grease" or "greases" or "FOG".

<u>Food Service Establishments</u> (FSE): Those establishments that engaged in activities of preparing, serving, or otherwise making available for consumption foodstuffs, and that use one or more of the following preparation activities: cooking by frying, baking, grilling, sautéing, rotisserie cooking, broiling (all methods), boiling, blanching, roasting, toasting, or poaching. Also included are infrared heating, searing, barbecuing, and any other food preparation activity that produces a hot, non-drinkable food product in or on a receptacle that requires washing. Those establishments that engage in the preparation of precooked and frozen food materials and meat cutting preparation and applicable to all "Food Service Establishments" that discharge wastewater containing grease to the City of Cayce Sanitary Sewer System including but not limited to the following: restaurants, grocery stores, meat markets, hotels, factory and office building cafeterias, public and private schools, hospitals, nursing homes, commercial day care centers, churches, and catering services.

<u>Food service establishment owner or owner</u> means in the case of individual FSE, the owner or proprietor of the food service operation. Where the FSE is a franchise operation, the owner of the franchise is the responsible person or entity. Where the FSE operation is owned by a corporation, the corporate representative is the responsible entity. Where two or more FSE share a common grease interceptor, the owner shall be the individual who owns or assumes control of the grease

interceptor or the property on which the grease interceptor is located.

<u>Gray water</u> means all of the liquid contained in a grease interceptor that lies below the floating grease layer and above the food solids layer.

<u>Grease waste hauler</u> (GWH) means a person and/or company who collect the contents of a grease interceptor or trap and transports it to an approved recycling or disposal facility. A grease hauler may also provide other services to a FSE related to grease interceptor maintenance.

<u>Grease Interceptor</u> is a large underground concrete vault; or other similar engineered material, located outside of an FSE designed to collect, contain, separate, or remove Grease from the waste stream while allowing the sub-straight waste or Gray Water to discharge to the wastewater collection system by gravity. Interceptors shall be in conformance with the provisions of the South Carolina Plumbing Code.

<u>Grease removal device</u> (GRD) refers generically to grease traps and grease interceptors.

<u>Grease trap</u> means a device located in a FSE usually under a sink designed to collect, contain or remove food wastes and grease from the waste stream while allowing the balance of the liquid waste to discharge to the wastewater collection system by gravity. Traps shall be in conformance with the provisions of this Policy and the South Carolina Plumbing Code.

<u>Superintendent</u>. The person designated by the City to supervise the operation of the POTW, including but not limited to, the implementation of the City's Grease Management Program (GMP), and who is charged with certain duties and responsibilities by this ordinance, or a duly authorized City representative.

Section 3: General

Requirements: Section 3.01:

Applicability:

The administrative, operational, and general requirements of this Policy are applicable to ALL FSE, new or existing. All FSE shall have grease removal and handling equipment approved by the City. Establishments whose grease removal device (GRD) is not in accordance with this standard shall be given a compliance schedule with a deadline not to exceed six (6) months from initial notification date to bring this equipment into compliance or install adequate equipment approved by the City. The Superintendent or designated city representative shall provide assistance in the facilitation of the FSE FOG training. In addition, educational materials shall be available through the GMP and City's website.

Section 3.02: Grease Removal Devices:

Plans and specifications for all new GRD's must be submitted to the City for reviewed and approved

by the City of Cayce Inspection Department and the Industrial Pretreatment Program prior to installation. Prior to placing any new GRD into operation, the Cayce Inspection Department and the Industrial Pretreatment Division shall inspect the GRD to assure compliance of approved GRD. No New FSE will be allowed to initiate operations until proper Grease Traps and/or Grease Interceptors, as appropriate, are installed by the FSE in accordance with this ordinance, and are approved by the City.

Grease traps and interceptors shall be provided for the proper handling of wastewater containing excessive amounts of fats, oil or greases, which will or tend to cause or contribute to sanitary sewer blockages and obstructions, except that such interceptors shall not be required for residential users. Food Service Establishments (FSE), commercial establishments and other industrial generating wastewater containing fats, oils or greases are subject to this ordinance. All existing and new FSEs must have Grease Traps and/or Grease Interceptors installed and approved by the City in accordance with the City FOG Control Policy. Grease Traps and/or Grease Interceptors shall be provided for all permitted food as defined by SCDHEC to include restaurants, cafeterias, snack bars, fast food restaurants or carry out food restaurants or any other areas such as service station or residential complex wash racks, automatic car washes or auto service areas, coin laundries, hospitals or other businesses which, in the opinion of the City, deem them necessary to for the proper handling of wastes generated by each business. All GRD units shall be of type and capacity approved by the Superintendent and shall be so located to be easily accessible for cleaning and inspection. Such GRD units shall be inspected, cleaned, and repaired regularly, as needed, by the user at their expense.

All Grease Traps and Interceptors shall be installed in accordance with the City's most recently adopted version of the International Plumbing Code. Grease Interceptors shall be installed in accordance with the requirements contained in all applicable local plumbing codes and must be installed by a licensed plumbing contractor. Wastewater from domestic facilities and other similar fixtures shall not be introduced into a Grease Trap or Grease Interceptor by the FSE under any circumstances.

- 1. <u>Design and Sizing</u>: Grease Traps and Grease Interceptors must be sized in accordance with the City of Cayce's FOG Control Policy formulae set forth in (A), (B), or (C) of the City of Cayce's *Grease Trap and Grease Interceptor Tank Sizing Guide* (Exhibit 1) unless the City approves sizing calculations signed and sealed by a registered professional engineer in accordance with the Fixture Unit Calculation Method set forth in (D) of the City of Cayce's *Grease Trap and Grease Interceptor Tank Sizing Guide* (Exhibit 1).
 - (a) All FSEs must submit, for each Grease Trap and Grease Interceptor, cut sheets, plans, and specifications. These documents must be submitted to the City prior to the FSE's installation of the Grease Trap and/or Grease Interceptor. An approval letter for each new Grease Trap or Interceptor must be issued by the City to the FSE prior to construction and/or installation of the Grease Trap and/or Grease Interceptor by the FSE. The construction and location criteria for Grease Interceptors must be in accordance with Environmental Protection Agency (EPA) Guidance Document, "On site Wastewater Treatment and Disposal Systems," Chapter 8.
- New FSEs Design and Sizing: New FSEs are any FSE's that are undergoing a change of use
 which necessitates the issuance of a new SCDHEC food service permit or are newly
 constructed and include existing FSE's expansion or renovation to provide food services. All New
 FSEs are required to install Grease Traps and/or Grease Interceptors, as appropriate, in compliance with

- 3. For Existing FSEs operating with GRD's that were installed prior to the effective date of this ordinance. These FSEs will generally be permitted to operate and maintain existing Grease Traps and/or Grease Interceptors provided they are in proper operating condition. If a Grease Trap or Grease Interceptor installed prior to the effective date of this ordinance does not allow for measurement and sampling to demonstrate that the Grease Trap or Grease Interceptor is in compliance with this regulation, the FSE must modify or replace the Grease Trap or Grease Interceptor as provided below within six months from the date of the revised sewer ordinance.
- 4. The City, in its sole discretion, may approve a proposed modification of an existing Grease Trap or Grease Interceptor which allows for measurement and sampling of the existing Grease Trap or Grease Interceptor to demonstrate that the Grease Trap or Grease Interceptor is in compliance with the City of Cayce's FOG Control Policy. Any proposed modification to meet this requirement must be submitted to the City for review no later than six months from the date of the revised FOG policy.
- 5. If not modified pursuant to paragraph (4) above, the existing Grease Trap or Grease Interceptor must be replaced with a new Grease Trap or Grease Interceptor, as appropriate, in compliance with the City of Cayce *Grease Trap and Interceptor Sizing Guide* (Exhibit 1).
- 6. Existing FSEs shall not connect new dishwashers, garbage grinders, or domestic sewer to any Grease Trap or Grease Interceptor. The City, in its sole discretion, may grant a variance to this requirement in circumstances in which the City determines that compliance with this requirement would be unduly burdensome or impractical due to physical condition or layout of the FSE.

A. Design and Structural Criteria for Exterior Grease Interceptors:

Grease Interceptors shall conform to the following general criteria when being designed and constructed:

- 1. <u>Location</u>: Any Grease Interceptor shall be located in the FSE's lateral sewer line between all fixtures which may introduce grease into the City's wastewater collection system. Grease Interceptors must be located as close to the source of the wastewater service line at the building as physically possible, while remaining accessible for cleaning and maintenance. All Grease Interceptors shall be located where they are easily accessible for inspection, cleaning, and maintenance.
 - (a) In cases where existing FSEs do not have a Grease Interceptor installed and where the installation of an outdoor Grease Interceptor is feasible in the City's sole discretion, the Grease Interceptor must be installed by the FSE in accordance with this Ordinance and must be approved by the City in writing in advance of installation.

In situations where it is not feasible for an existing FSE to install an underground Grease Interceptor, the FSE is required to install adequate and approved Grease Traps for use on individual fixtures, including, but not limited to: pot sinks, mop sinks, pre-rinse sinks, wok ovens, floor drains, and any other drains where the potential for introduction of Grease exists. In such cases, Grease Traps will be considered adequate and will be approved by

the City only if a flow control device is placed on the inlet that prevent overloading and a sample port is placed on the outlet of each Grease Trap

- 2. <u>Design and Sizing</u>: The design and size of all interceptor units must comply with Exhibit 1 & 3 of this ordinance. The size, type and location of each grease interceptor shall be approved by the City based on plans and specifications submitted to by the user. FSEs must comply with standard details GR #1 (Exhibit 3) and GR #2 (Exhibit 3) for standard Grease Interceptor installation. Grease Interceptors may be installed in series, if volume required is more than 1500 gallons. When installed in series, the first tank shall not have a baffle or center tee (see standard detail GR #2, Exhibit 3). Design is the responsibility of the user/owner/developer. No device shall be installed prior to written approval from the City.
- 3. No Grease Interceptor less than 1,000 gallons total capacity will be approved for installation by the City unless acceptable engineering calculations sealed by a professional engineer registered in the state of South Carolina and demonstrating that a smaller size has satisfactory capacity are provided by the FSE Owner to the City and are approved by the City. The City retains sole discretion to approve or deny approval of a Grease Interceptor less than 1,000 gallons total capacity in all circumstances, even in the event such engineering calculations are provided.
- 4. A minimum hydraulic retention time of 24 minutes at actual peak flow between the influent and effluent baffles, with 25 percent of the total volume of the grease interceptor being allowed for any food-derived solids to settle or accumulate and floatable grease derived materials to rise and accumulate.
- 5. Grease Interceptors shall be precast or cast in place minimum 4000 psi concrete vaults. The City, in its sole discretion, may allow the use of other materials upon written request for approval prior to installation. Grease Interceptors shall contain the following:
 - (a) Two-way cleanouts must be provided by the FSEs on the inlet and outlet lines of the Grease Interceptor.
 - (b) Open top inlet tee must extend to 2/3 the depth of the fluid level of the vault.
 - (c) The vault shall have a baffle wall with a minimum 3" air gap for venting at the top and an open top tee extending to ??" from the bottom of the vault.
 - (d) Open top outlet tee must extend to ??" from the bottom of the vault.
 - (e) Anti-flotation design with proper base course and compacted sub-grade are to be used to prevent settling where conditions warrant.
- 6. Access Manholes shall be provided by the FSE directly above all three tees for inspection and maintenance. Access manholes shall have an installed diameter of 24 inches, a maximum weight of 50 pounds, and shall be provided over each chamber, interior baffle wall, and each sanitary tee. The access penetrations, commonly referred to as "risers" into the grease interceptor shall also be, at a minimum, 24 inches in diameter. The access manholes shall

extend at least to finished grade and be designed and maintained to prevent water inflow or infiltration. The manholes shall also have readily removable covers to facilitate inspection, grease removal, and wastewater sampling activities. Manhole frames and covers shall conform to the following criteria:

- (a) An access opening shall be provided by the FSE above the inlet, baffle wall, and outlet tees and shall be easily removable by one person. A minimum 6" diameter traffic rated clean out with a concrete collar extending down through the vault top may be provided by the FSE above the baffle wall tee in lieu of an access manhole.
- (b) Manhole frames and covers shall be grey iron castings, ASTM A48, Class 30 Iron with all bearing surfaces machined.
- (c) Acceptable manufacturer: US Foundry Model 680; and provide HS-20 rated frame and cover where applicable.

B. Interior Grease Traps:

Grease Traps shall conform to the following general criteria when being designed and constructed:

- 1. <u>Location</u>: All Grease Traps shall be located between all fixtures in the FSE and the interior piping which is connected to City's wastewater collection system. The FSE is required to install adequate and approved Grease Traps for use on individual fixtures, including, but not limited to: pot sinks, mop sinks, pre-rinse sinks, ovens, floor drains, and any other drains where the potential for introduction of grease
 - exists. Each grease trap shall be so installed and connected that it will be readily accessible for cleaning and inspection at all times.
 - (a) In cases where circumstances will not allow for the installation of an outdoor in ground grease interceptor and only with the approval of the City, the FSE may be allowed to install an indoor GRD. The FSE shall submit plans and specs to include unusual circumstances warranting a GRD. The application shall (i) describe the FSE (i.e., hours of operation, type of food sold and average number of customers served on a weekly basis); (ii) describe the food preparation process for the business; (iii) the type of proposed interior grease trap; and (iv) processes and procedures used to ensure FOG does not leave the grease trap into the City's wastewater system. If approved, the GRD variance shall be in force until there is either (i) a change in ownership of the FSE; (ii) evidence demonstrates that a GRD is no longer necessary; (iii) evidence that the GRD is not keeping FOG out of the City's wastewater system; and/or (iv) remodeling of the kitchen or facility.
- (2) <u>Design and Sizing</u>: Sizing of any Grease Trap must be in accordance with the City of Cayce's *Grease Trap and Interceptor Sizing Guide* (attached as Exhibit 1). The minimum size of grease traps shall be based on the maximum rate of flow of all fixtures discharging into the grease trap multiplied by a retention factor of 1.5 minutes. No grease trap shall be installed with an approved rate of flow less than 20 gallons per minute or a retention capacity of less than 40 lbs. No garbage disposal or dishwasher shall be connected to or discharged into any grease trap.
- (3) Grease traps shall be constructed of durable materials satisfactory to the Superintendent and shall have a full size gas tight cover, which can be readily removed. Each grease trap shall have a water seal of not less than 2" in depth or the diameter of its outlet, whichever is greater. No single in-line grease trap shall serve more than two separate fixtures. Grease traps shall be installed and vented in accordance with the International Plumbing Code.
- (4) Cleaning and maintenance should generally be performed in accordance with the Grease Trap

manufacturer's recommendations. However, despite this provision, cleaning and maintenance must be performed as often as and in the manner necessary to achieve full compliance with the City of Cayce FOG Control Policy, even if such cleaning and maintenance exceeds that recommended by the manufacturer. At a minimum, Grease Traps must be cleaned on a quarterly basis but as often as necessary to insure FOG are not discharged into the collection system.

C. Inspection and Operational Approval:

The FSE must notify the City 72 hours in advance, during the normal work week, when the Grease Trap and/or Grease Interceptor is ready for final inspection by the City. The City must give final approval prior to the FSE covering any Grease Interceptor. Where the City denies approval of the Grease Interceptor for a new FSE, the City may elect to submit a request to the appropriate building official requesting that certificates of occupancy be withheld until the Grease Interceptor is constructed in accordance with this specification and is approved by the City.

Section 4: Alternative grease removal devices or technologies:

Alternative devices and technologies such as automatic grease removal systems shall be subject to written approval by the Superintendent prior to installation. Approval of the device shall be based on demonstrated (proven) removal efficiencies and reliability of operation. The City may approve these types of devices depending on manufacturer's specifications on a case by case basis. The FSE may be required to furnish analytical data demonstrating that grease discharge concentrations to the City's wastewater collection system will not exceed the established limitations.

The use of biological or other additives as a grease degradation or conditioning agent is not permissible for use within the Cayce's collection system. Flushing a GRD with hot water, or the use of chemicals or other agents to dissolve or emulsify grease and allow it to flow into the wastewater treatment system, is a violation of this Policy

Section 5: Records Maintenance, Reporting and Retention:

A. Record Keeping:

Each FSE shall maintain records of all Grease trap or interceptor maintenance. To including the date and time of the maintenance, details of any repairs required and dates of repair completion, cleaning and any other records pertaining to the grease trap or interceptor. These records shall be made available for review upon request by the Superintendent. Each FSE shall also maintain a file on-site which contains the following information:

- 1. The (as-built) drawings of the plumbing system, if available. If as-built drawings are not available, other drawings of sufficient detail to depict the plumbing layout of the FSE.
- 2. Receipts/reports from grease waste haulers and plumbing repairs.
- 3. Grease waste hauler information.

The FSE shall maintain a written record of grease trap or interceptor maintenance and cleaning for three (3) years. All such records will be available for inspection by the City at all times. These records

shall include but not be limited to (See Exhibit 4):

- 1. FSE name and physical location
- 2. Date and time of grease trap or interceptor service
- 3. Name and signature of grease waste hauler performing said service
- 4. Established service frequency and type of service.
- 5. Number and size of each grease interceptor serviced at FSE location
- 6. Total volume of waste removed from each grease trap or interceptor
- 7. Destination of removed wastes, food solids, and wastewater disposal
- 8. Signature and date of FSE personnel confirming service completion
- 9. Such other information as required by Superintendent.

The file shall be available at all times for inspection and review by the Superintendent. The failure to maintain complete records or to provide such records to the Superintendent upon request constitutes a violation of this Policy.

Annually the city will mail each FSE a request to submit copies of all grease trap and/or interceptor cleaning to the City for review. FSE will have 30 days to provide copies of these records.

Section 6: Pumping/Cleaning and Maintenance:

Interceptor pumping frequency: Each FSE shall have its grease trap(s) and/or interceptor(s) pumped at a **minimum frequency of once every 90 days** but as often as necessary to insure FOG is not discharged in to the collection system. In addition to required pumping, each FSE shall determine additional frequency at which its grease interceptor(s) shall be pumped according to the following criteria:

- 1. When the floatable grease layer exceeds six (6) inches in depth as measured by an approved dipping method, or;
- 2. When the settleable solids layer exceeds eight (8) inches in depth as measured by an approved dipping method, or;
- 3. When the total volume of captured grease and solid material displaces more than 25 percent of the capacity of the interceptor as calculated using an approved dipping method, or;
- 4. When the interceptor is not retaining/capturing oils and greases.

Each FSE shall be responsible for the costs of installing, inspecting, pumping, cleaning, and maintaining its grease trap or interceptor. Since the FSE is liable for the condition of their

pretreatment devices it is recommended they witness all cleaning/maintenance activities to verify that the grease trap or interceptor is being fully cleaned and properly maintained. All FSE that have grease traps or interceptors shall utilize a GWH who has been permitted by the SCDHEC for pumping services. Pumping services shall include the complete removal of all contents, including floating materials, wastewater, and bottom sludges and solids from the trap and/or interceptor. Recycling of wastewater is not permissible.

The following are the pumping practices required of registered grease waste haulers (Steps 1-12):

- **Step 1.** Record reading using appropriate dipping method (sludge judge reading, etc.).
- **Step 2.** Skim the entire grease cap and debris from the top of the Grease trap or Interceptor. The trap or interceptor may need to be agitated slightly to loosen the grease cap.
- **Step 3.** Place the vacuum tube all the way into the Grease Interceptor to withdraw remaining solids from the bottom.
- **Step 4.** Vacuum water out of the Grease trap or Interceptor.
- **Step 5.** Clean the sides and bottom of the Grease trap or Interceptor. This may be done by "back flowing" the water from the pump truck or by using an alternative water source to hose down the trap or interceptor. Make sure the Grease trap or Interceptor is completely clean.
- **Step 6.** Vacuum the remaining water out of the Grease trap or Interceptor.
- **Step 7.** Check that the sanitary "Tees" on the inlet and outlet sides of the Grease Interceptor are not clogged, loose, or missing.
- **Step 8.** Verify that the baffle is secure and in place.
- **Step 9.** Inspect the Grease trap or Interceptor for any cracks or other defects.
- **Step 10.** Allow FSE contact to inspect the cleaning for their approval. It is the responsibility of each FSE to inspect its Grease trap or Interceptor during and after the pumping and cleaning procedure to ensure that the Grease trap or Interceptor is properly cleaned out and that the structure is sound and all fittings and fixtures inside the Grease trap or Interceptor are in working condition and are functioning properly.
- **Step 11.** Check that lids are securely and properly seated after completion of pumping.
- **Step 12.** Provide a copy of the waste hauler manifest to the FSE
- **A. Disposal Requirements:** Wastes removed from each grease trap or interceptor shall be disposed of at a facility permitted by the SCDHEC to receive such wastes. Neither grease nor solid materials removed from interceptors shall be returned to any grease trap, interceptor, private sewer line or to any portion of the City's wastewater collection system or water reclamation facilities.

Section 7: Right of Entry:

The City will perform a physical inspection of each FSE facility on a routine schedule to insure compliance with all conditions specified in this FOG policy.

- 1. The City (or its designee) shall have right-of-entry on or upon the property of any FSE subject to this Policy for the purpose of inspection, determination of compliance, or enforcement action. The City shall be provided ready access to all parts of the premises for the purposes of inspection, monitoring, sampling, ability to establish limits, inventory, records examination and copying, and the performance of any other duties which the City reasonably deems necessary to determine compliance with this Policy.
- 2. Where security measures are in place which require proper identification and clearance before entry into the premises, the property owner or commercial enterprise shall make necessary arrangements with its security guards so that, upon presentation of proper credentials, the City will be permitted to enter without delay for the purposes of inspecting the site and investigating any complaints or alleged violations of this Policy.
- 3. The City shall have the right to set up on the property such devices as are necessary to conduct sampling and/or metering and shall have the power to require written statements, certificates, certifications, or the filing of reports relating to complaints or alleged violations of this Policy.
- 4. All grease removal devices (GRD) shall be easily accessible and shall not be obstructed by landscaping, parked cars, or other obstructions. Any temporary or permanent obstruction to safe and easy access to the areas to be inspected and/or monitored shall be removed promptly by the responsible party at the written or verbal request of the City. The costs of clearing such access shall be borne by the responsible party.

Section 8: Grease Waste Hauler (GWH)

Prior to a GWH providing grease trap or interceptor cleaning service within the Cayce wastewater collection system the hauler must have a current SCDHEC approved permit number. They must also have a current City of Cayce business license if performing this service within the city limits.

A. Spill Reporting:

Any accident, spill, or other discharge of grease or gray water which occurs within the City shall be reported to the City by the waste hauler as soon as possible but not longer than 24 hours after the incident. The waste hauler shall comply with all procedures contained in federal, state and local regulations. The waste hauler shall be responsible for all clean-up procedures and costs.

B. Record Keeping:

Grease waste haulers shall retain and make available for inspection and copying, all records related to grease interceptor pumping and grease disposal from businesses located in the wastewater collection service area. Records are required for each trip to a disposal facility to include the vehicle, date, amount of waste and name of disposal or recycling facilities. These records shall remain available for a period of at least three years. The failure to provide information to the City within ten days of

a written request is a violation of this Policy and may result in revocation license. The City may require additional record keeping and reporting, as necessary, to ensure compliance with the terms of this Policy. Repeated failure of a GWH to submit required or requested reports in a timely manner or the repeated submission of incomplete reports will result in the City revoking the haulers business license, the hauler being banned from providing grease removal service within the Cayce utility service area and reporting of suspect actions to SCDHEC for permit compliance evaluation.

Section 9: Enforcement

A. Administrative Enforcement Activities:

The City is authorized, but not limited to taking the following actions in response to noncompliance with the Fats, Oils and Grease Policy:

- 1. Written Notice of Deficiency [NOD]
- 2. Written Notice to Correct [NTC]
- 3. Notice of Violation [NOV]
- 4. Enforcement Meeting
 - (a) Administrative Order [AO]
 - (b) Initiation of Self-Monitoring
 - (c) Employee Training Requirement
 - (d) Consent Order (may contain Schedule of Compliance)
- 5. Administrative Fines and Re-inspection Fees. Penalties per EPA's Streamlining Rule, SC Reg. 61-9 and as set forth and established by City Council on Appendix B to the City Code
- 6. Payment of Remediation/Clean-Up Costs and/or Cost Recovery
- 7. Termination of (Sewer and/or Water) Service [TOS]

B. Recovery of costs:

When a discharge of waste causes an obstruction, damage or any other impairment to the facilities, or any expense of whatever character or nature to the City, the Superintendent may assess the expenses incurred by the City to clear the obstruction, repair damage to the facility, and any other expenses or damage of any kind or nature suffered by the City. The Superintendent shall file a claim with the FSE or any other person or entity causing such damages seeking reimbursement for any and all expenses or damages suffered by the City. If the claim is ignored or denied, the Superintendent shall notify the City Attorney to take such measures as shall be appropriate to recover any expense or to correct other damages suffered by the City.

C. Remedies nonexclusive:

The remedies provided for in this Policy are not exclusive. The City may take any, all, or any combination of these actions against a person violating this Policy. Enforcement of violations will

generally be in accordance with this Policy; however, the City may take other action against any person when the circumstances warrant. Further, the City is empowered to take more than one enforcement action against any person in violation of this Policy.

APPENDIX B - FEE SCHEDULE

Chapter 40. Utilities

Article III. Sewer System Division 2. Sewer Rates, Tap Fees § 40-119. Fats, oils and grease:

FSE First reinspection (violation) fee\$250.00 FSE Successive reinspection (violation) fee\$500.00 Septic Waste Disposal fee \$0.09 per gallon Grease Waste Disposal fee \$0.16 per gallon Hauler Truck Cleaning fee \$100.00

STATE OF SOUTH CAROLINA	ORDINANCE 2019-17	
COUNTY OF LEXINGTON	Amending City Code Section 4("Prohibited Discharges") of Ar	ticle
CITY OF CAYCE	 III ("Sewer System") of Chapte ("Utilities") 	r 40
Discharges") of Article III ("Sew	es to amend City Code section 40-92 (ver System") of Chapter 40 ("Utilities") is Fats, Oils and Grease Control Policy,	
NOW, THEREFORE, BE IT Cayce, in Council, duly assembled,	ORDAINED by the Mayor and Council of as follows:	the City of
Section 40-92 ("Prohibited D 40 ("Utilities") of the Cayce City Cod	Discharges") of Article III ("Sewer System") de is hereby amended to read:	of Chapter
escape into the sanitary sewer s would damage or injuriously aff shall include, but not be limite chemicals, clothing, rags, remna- affect, injure, or retard the bio material. By reference, prohibite ordinance latest edition. Non- ordinance, or the Fats, Oils and G	in, allow to escape into, or cause or system of the city any material or substituted the operation of the system, which ed to, petroleum products, fats, oils, nts, gases, refuse or any other substance logical treatment system receiving substances are also cited in the scompliance with this section, the screase (FOG) policy also authorizes the ced in the enforcement sections of the scream in the scream in the enforcement sections of the scream in the enforcement sections of the scream in	tance that deposits greases, se that will ich waste sewer use ewer use sity to take
This Ordinance shall be effect	ctive from the date of second and final read	ding.
DONE IN MEETING DULY	ASSEMBLED, this day of	2019.
	Elise Partin, Mayor	
Attest:		
Mendy Corder, CMC, Municipal Cle	 ∍rk	
First Reading:		

Second Reading and A	Adoption:	
Approved as to form:	Danny C. Crowe, City Attorney	

Attachment A

Sec. 40-92. Prohibited discharges.

No person shall deposit in, allow to escape into, or cause or permit to escape into the sanitary sewer system of the city any material or substance that would damage or injuriously affect the operation of the system, which deposits shall include, but not be limited to, petroleum products, fats, oils, greases, chemicals, clothing, rags, remnants, gases, refuse or any other substance that will affect, injure, or retard the biological treatment system receiving such waste material. By reference, prohibited substances are also cited in the sewer use ordinance latest edition. Non-compliance with this section, the Sewer Use Ordinance, or the FOG policy authorizes the City to take enforcement actions as described in the enforcement section of the FOG policy.

Memorandum

To: Mayor and Council

From: Tracy Hegler, City Manager

Carroll Williamson, Planning and Development Director

Date: September 19, 2019

Subject: First Reading of an Ordinance Amending Zoning Ordinance Article 2 Definitions,

Article 8 Table 8-B Number, Dimension, and Location of Signs Requiring a Permit and Article 8 Table 8-B Table Notes to Add Language Concerning Digital Display

Signs

Issue

Council approval is requested for the First Reading of an Ordinance to amend the Zoning Ordinance to better regulate digital display signs.

Discussion

Digital display signs in the City of Cayce are currently permitted but have restrictions that are not necessarily reflective of the current capabilities of these types of signs. This amendment retains the restriction of no animation or scrolling messages on digital signs. However, it permits multiple colors and characters instead of just letters. Additionally, in accordance with International Sign Association's recommendations, the duration for each image shall remain for a minimum of 8 seconds, the transition between images shall be less than one second, and the brightness of the sign cannot be greater than 0.3 foot candles relative to the surrounding light. Each sign will be required to have a sensor that adjusts its brightness based on ambient light conditions. This amendment makes no changes to size or height of digital signs.

The Planning Commission voted on amending the Zoning Ordinance pertaining to digital display signs at the Planning Commission meeting on September 16, 2019.

Recommendation

The Planning Commission recommends approval of the First Reading of an Ordinance Amending the Zoning Ordinance pertaining to digital display signs.

STATE OF SOUTH CAROLINA)	,		ORDINANCE 20	
COUNTY OF LEXINGTON)	2 Definition	ons, Article 8 Ta	
CITY OF CAYCE	,	Signs Red Table 8-B	Dimension, and quiring a Permit Table Notes to ng Digital Displ	t and Article 8 Add Language
WHEREAS , Council has the Zoning Ordinance Article 2 Do of Signs Requiring A Permit and and	efinitions, Ar	ticle 8 Table	8-B Number, D	imension, and Location
WHEREAS, the Planning request to receive comments from		_	ularly scheduled	d public hearing on this
WHEREAS, the Planning comments and vote on recomm Number, Dimension, and Location Notes of the Cayce Zoning Ording shown on the attached documents	ending the a on of Signs nance and de	amended Ai Requiring A	ticle 2 Definition Permit and Art	ns, Article 8 Table 8-B ticle 8 Table 8-B Table
NOW, THEREFORE, BE in Council, duly assembled, that A Location of Signs Requiring A Peshown on the attached document	Article 2 Defirermit and Arti	nitions, Artic	le 8 Table 8-B N	umber, Dimension, and
This Ordinance shall be e	ffective from	the date of	second reading	approval by Council.
DONE IN MEETING DUL	Y ASSEMBI	LED, this	day of	2019.
		Elise Pa	artin, Mayor	
Attest:				
Mendy Corder, CMC, Municipal (Clerk			
First Reading:				
Second Reading and Adoption: _				
Approved as to form:		it., Attown		
Danny (C. Crowe, Ci	ily Attorney		

<u>Sign</u> - Any object, device, display or structure, or part thereof, situated outdoors or indoors, which is used to advertise, identify, display, direct or attract attention to an object, person, institution, organization, business, product, service, event or location by any means, including words, letters, figures, design, symbols, fixtures, colors, illumination or projected images.

<u>Sign, A-Frame</u> - A freestanding movable sign, not secured or attached to the ground or any building or structure, composed of a sign panel and supporting structure or one or more panels which form both the structure and sign face. (Also known as a "Sandwich Board")

<u>Sign, Abandoned</u> - A sign structure not containing a sign for 120 continuous days or a sign not in use for 120 continuous days, or a sign advertising a business no longer occupying the site on which the sign exists or to which it refers.

<u>Sign, Animated</u> - Any sign that uses movement or change of lighting to depict action or creates a special effect or scene.

<u>Sign, Awning, Canopy or Marquee</u> - A sign that is mounted or painted on or attached to an awning, canopy or marquee.

<u>Sign, Banner</u> - Any sign applied to or constructed of paper, plastic or fabric of any kind with or without frame, and with or without design or lettering, used to decorate or attract attention to a location, object, institution, product service or business. Flags of nations, states or political subdivisions, shall not be considered banners.

<u>Sign, Bench</u> - A sign located on any part of the surface of a bench or seat placed on or adjacent to a public right-of-way.

Sign, Building - Any sign attached to any part of a building.

<u>Sign, Changeable Copy</u> - A sign or portion thereof with characters, letters or illustrations that can be changed or rearranged without altering the face or the surface of the sign. A <u>sign on which the message changes more than eight times per day shall be considered an animated sign and not a changeable copy sign for purposes of this Ordinance.</u> A sign on which the only copy that changes is an electronic or mechanical indication of time or temperature shall be considered a "time and temperature" portion of a sign and not a changeable copy sign for purposes of this Ordinance.

<u>Sign, Digital Reader Display — The portion of a sign message made up of internally illuminated components capable of changing the message periodically. Digital displays may include but are not limited to LCD, LED, or plasma displays. Commonly known as Outdoor LED Message Display Boards shall be limited to one-color static (fixed). There shall be no scrolling capabilities and copy change or message can change no more than eight times in one calendar day.</u>

Characters or digits shall be no more than eight (8) inches in height. Allowable only as a percentage of total allowable signage for freestanding signs.

<u>Sign</u>, <u>Face</u> - The area or display surface used for the message.

Sign, Free-Standing - Any non-movable sign not affixed to a building.

<u>Sign, Identification</u> - A sign giving the nature, logo, trademark, or other identifying symbol; address; or any combination of the name, symbol, and address of a building, business, development, or establishment on the premises where it is located.

<u>Sign, Incidental</u> - A sign, generally informational, that has a purpose secondary to the use of the zone lot on which it is located, such as "no parking", "entrance", "loading only", "telephone", and other similar directives. No sign with a commercial message legible from a position off the zone lot on which the sign is located shall be considered incidental.

<u>Sign, Inflatable</u> - A sign that requires air, whether contained or blown, to keep and maintain its shape, or which causes an object to flutter.

<u>Sign, Off-Premise</u> - A sign which directs attention to a business, commodity, service, or entertainment conducted, sold, or offered at a location other than the premises on which the sign is located.

<u>Sign, Pennant</u> - Any light-weight plastic, fabric or other material, whether or not containing a message of any kind suspended from a rope, wire or string, usually in series, designed to move in the wind.

<u>Sign, Permanent</u> - A sign attached to a building, structure or the ground in some manner and made of materials intended for more than short term use.

<u>Sign, Political</u> - A temporary sign announcing or supporting political candidates or issues in connection with any national, state or local election.

<u>Sign, Portable</u> - A sign that may be moved from one location to another, is not permanently affixed to the ground, and is differentiated from a movable sign in that it may be equipped for transportation by motor vehicle or other mechanical means. Trailer signs are considered to be portable signs.

<u>Sign, Posters</u> - See Also Banner. For the purposes of this ordinance, a Poster is considered a banner.

<u>Sign</u>, <u>Projecting</u> - A sign that is wholly or partly dependent upon a building for support and which projects more than 12 inches from such building.

Table 8-B			
Number, Dimension, and Location of Signs Requiring A Permit			
By Zoning District			

	All Residential	C-1	C-2	C-3	C-4	M -1	M-2	D-1	INS(B)
F	Zones								
Freestanding Signs									
Number Permitted (E)									
Per Lot									
Off-Premise	N	N	N	N	N	NA	NA	N	N
(Billboards)	IN	IN	IN	IN	IN	INA	INA		IN
On-Premise	1(A)	1	1	1	1	1	1	1(A)	1
Per feet of St. Frontage									
Off-Premise	N	N	N	Ν	N	(C)	(C)	N	N
On-Premise	NA	NA	NA	NA	NA	(D)	(D)	NA	NA
Maximum Sign Area									
(s.f.)									
Off-Premise	NA	NA	NA	NA	NA	672	672	NA	NA
On-Premise	20	75(H)	75(H)	75(H)	75(H)	75(H)	75(H)	20	20
Sign, Digital				60%	60%	60%			
<u>Display</u> Reader s.f.	NA	NA	NA	(<u>L)</u>	(L)	(<u>L)</u>	NA	NA	NA
max. allow. area				<u>\L/</u>	<u>\L/</u>	<u>(L)</u>			
Minimum Setback									
from property line									
Billboards	50'	10'	10'	10'	10'	10'	10'	10'	10'
Other	5'	5'	5'	5'	5'	5'	5'	5'	5'
Maximum Height	12'	20'	20'	7'(I)	20'(J)	40'(F)	40'(F)	20'	12'
Building Signs									
Number Permitted	1	6	6	6	6	6	6	1	1
Maximum Sign Area	2	200	200	200	200	200	200	12	12
(s.f.)	_	200	200	200	200	200	200	12	12
Maximum Wall Area	NA	15	15	15	15	15	15	NA	NA
(Primary Wall) (%) (G)	14/7	.0	.0		.0	- 0	10	14/1	14/1
Maximum Wall Area									
(Secondary Wall) (%)	NA	5	5	5	5	5	5	NA	NA
				Sec. C	ootion 0	6			
Temporary Signs (2)				See S	ection 8	.0			

Table Notes:

NA = Not Applicable

N = Not Allowed

s.f. = Square Feet

- A. One use identification sign not exceeding 20 s.f. is permitted for each entrance of a subdivision or residential project.
- B. This column does not represent a zoning district. It applies to institutional and other non-residential uses permitted under the Zoning Ordinance in residential districts, i.e. churches, schools, parks, etc.

- C. Minimum distance between billboards shall be 2000' measured on the same side of the street along the centerline of the street from which the billboard is viewed. When meeting this requirement, a billboard may be located on the same lot or parcel with another billboard or off-premise sign.
- D. One per lot or one for each 500 linear feet of street frontage, whichever is less.
- E. Lots fronting on two or more streets are allowed one additional sign for each street frontage, but signage cannot be accumulated and used on one street in excess of that allowed for lots with only one street frontage.
- F. Maximum height of billboards may extend to 75', provided the sign structure is setback from the nearest property line one additional foot for each additional three feet in height over 40 feet.
- G. Wall area is calculated by multiplying the linear feet of the wall, in which the sign is to be placed, by a standard 12 feet height, as established by this ordinance.
- H. Sign area may be increased in accord with common signage plan requirements, where applicable, Section 8.4.
- I. Sign height may be increased in accord with common signage plan requirements, where applicable, Section 8.4.
- J. The maximum sign height may be increased in the C-4 District one foot for each linear foot of street frontage, not to exceed 35 feet in height.
- K. Corner lots fronting on two or more streets may be allowed more than one primary frontage and be allowed 15% maximum sign area for all walls having primary frontage.
- K.L. The messages on digital display signs shall have a minimum duration of eight (8)8 seconds. No scrolling is permitted and message transition shall be under one second. The difference between the off and solid-message measurements shall not exceed 0.3 footcandles. All permitted digital display signs shall be equipped with a sensor or other device that automatically determines the ambient illumination and programmed to automatically dim according to ambient light conditions.

Memorandum

To: Mayor and Council

From: Tracy Hegler, City Manager

Carroll Williamson, Planning and Development Director

Date: September 27, 2019

Subject: Memorandum of Understanding and Agreement between the City of Cayce and

the Town of Springdale for Building Official Services

Issue

Council approval is needed for the renewal of the Memorandum of Understanding (MOU) and Agreement for the City of Cayce to provide commercial and residential building plan reviews and inspections to the Town of Springdale at the rate of \$1,000 per month until January 1, 2022.

Discussion

On December 18, 2014, Cayce City Council and Springdale City Council signed a Memorandum of Understanding and Agreement for Cayce's Building Official to provide plan review and inspection services for \$1,000 per month. This agreement was renewed in June 2018 and expires on January 1, 2020.

The proposed agreement is identical as the current agreement but extends the agreement until January 1, 2022.

In order for this Agreement to go into effect, Springdale Town Council must also approve it. It will be on their agenda at their next meeting on October 1, 2019.

Recommendation

Staff recommends approval of the Memorandum of Understanding and Agreement between the City of Cayce and the Town of Springdale for Building Official services.

STATE OF SOUTH CAROLINA) MEMORANDUM OF UNDERSTANDING AND AGREEMENT BETWEEN CITY OF COUNTY OF LEXINGTON) CAYCE AND TOWN OF SPRINGDALE

THIS MEMORANDUM OF UNDERSTANDING AND AGREEMENT is made and entered into this _____ day of ______, 2019, by and between the City of Cayce, South Carolina ("Cayce") and the Town of Springdale, South Carolina ("Springdale") with approval of their governing bodies.

WHEREAS, Springdale desires to contract with Cayce for Cayce to provide required building code inspection and plan review of commercial and residential buildings for Springdale for the purpose of providing code compliance for construction projects; and

WHEREAS, Springdale and Cayce recognize the positive impact this contract will have in maintaining continuity of essential services through inspections and plan review on all commercial and residential projects; and

WHEREAS, Springdale agrees to reimburse Cayce for the cost of inspections and plan reviews as indicated below,

NOW, THEREFORE, in consideration of the services and agreement described herein, the parties hereto agree as follows:

- 1. Springdale agrees to compensate Cayce for provision of services as follows:
 - a. Springdale agrees to pay, directly to Cayce, a flat payment rate of \$1,000 per month (\$12,000 per year). Springdale agrees to pay this amount no later than the 15th of each month during the life of this agreement.
- 2. Cayce will provide building code inspections and plan reviews of commercial and residential buildings for Springdale, as follows:
 - a. The Licensed General Contractor or Residential Home Builder (hereinafter "contractor") shall obtain approval(s) and all related permits from Springdale for a commercial or residential building located, or to be located, within the boundaries of Springdale.
 - b. The contractor for a project shall submit plans for review and pay fees to Springdale. Springdale shall deliver said plans to the Cayce Building Official (hereinafter "Official"). Building plans may be submitted to the Official prior to Springdale's approval(s) in order to expedite the permitting process if desired by the contractor and approved by Springdale.
- 3. Contractors shall call in all inspection requests to Springdale and Springdale shall keep a daily log of all inspection requests and inspections conducted.

- 4. Springdale and Cayce agree that services for inspections and plan review will be handled by licensed inspectors and plans examiners, as required by the South Carolina Department of Labor, Licensing and Regulation.
- 5. The Building Official of Cayce shall interpret provisions of the applicable Building Code(s). Such interpretations may be appealed to the City of Cayce Code Board of Appeals. Fees for an appeal shall be as set forth by Cayce ordinance. In the event of an appeal, the City of Cayce Building Official will testify as to code requirements.
- 6. Springdale and its successors and assigns do hereby remise, release, acquit, and forever discharge Cayce, its employees, agents, successors, and assigns, from any and all actions, causes of action, claims, demands, damages, costs, expenses, third party actions, suits at law or for indemnity of whatever nature, and all consequential damage on account of, or in any way arising from, the services rendered under this Agreement. Springdale further agrees to hold harmless and indemnify Cayce for any and all losses, demands, damages, costs, claims, suits, and other liability arising from the services rendered by Cayce and by its employees and agents acting within the scope of their duties for Cayce or under this Agreement.
- 7. This Agreement shall continue in force until January 1, 2022, unless terminated sooner, in writing, by either party, or upon Springdale's employment of its own Building Official, or upon Cayce's inability to provide said inspection services. Either party may terminate this Agreement only by delivering written notice to the other party at least 60 days prior to the effective date of such termination. This agreement may be extended by written amendment approved by the Springdale Council and the Cayce Council.

IN WITNESS WHEREOF, WE, THE UNDERSIGNED, have this day, 2019, set our hands and seals hereon.				
City of Cayce	WITNESSES:			
City Manager				
Town of Springdale	WITNESSES:			
 Mayor				

Memorandum

To: Mayor and Council

From: Tracy Hegler, City Manager

Carroll Williamson, Planning and Development Director

Date: September 27, 2019

Subject: Bid Acceptance for Art Lot Construction

Issue

Council's approval is needed in order to accept the bid from 3D Dirtworks for the construction of the City's Art Lot located at 1908 Street.

Discussion

On January 30, 2019, the City was awarded a Connected Communities grant from the Central Carolina Community Foundation to construct an Art Lot at 1908 State Street, located in the City's newly established Arts District.

After a public input session on August 28th, a request for bids went out on September 12th. On September 26th, the City received 4 bids.

Construction Firm	Amount
JF Contractors	\$135,720.00
3D Dirtworks	\$98,800.00
Southern Vistas	\$99,912.00
AOS Specialty Contractors	\$167,038.00

City staff and the project designer Landplan Group South selected 3D Dirtworks of Sumter, SC with a bid of \$98,800.00. This was the lowest bid submitted.

The Art Lot will consist of a K-9 Memorial funded in partnership with the Cayce Public Safety Foundation, two (2) permanent public art pieces, a shelter and grassy meeting area, places to create new art, and seven (7) parking spaces to support the growing businesses in this area of the Arts District.

The intent is for the Art Lot to be completed by late January 2020.



MEMO

To: Tracy Hegler, City Manager

City of Cayce

Re: Recommendation of Award

Cayce Art Lot Cayce, SC

Date: 09/26/2019

From: Charles Howell Jr., RLA, ASLA

On Thursday, September 26, 2019, four (4) bids were received for the above referenced project. A tabulation of these bids is attached for your information and review. *The City has elected to utilize <u>Alternate 1</u> in lieu of item 13a (Shade Sails). The apparent low bid accounts for that adjustment. We have reviewed the low bidder's references and are satisfied with the responses. We have contacted LLR, and the BBB and the low bidder has no known issues with either agency.

The LandPlan Group South recommends award to the lowest, responsive bidder, 3D Dirtworks of Sumter, SC, in the base bid with alternate amount of *\$98,800.00.

Please contact me if you have questions or if you need additional information.

Memorandum

To: Mayor and Council

From: Tracy Hegler, City Manager

Jim Crosland, Assistant City Manager

Date: September 25, 2019

Subject: Approval of a Proposal for Cayce Arts District Design and Management

Issue

Council's approval is needed in order to accept a proposal from Will Bryan Design to manage and design a highly visible Arts District on State and Frink Streets.

Discussion

The City of Cayce is in the process of implementing its vision for an Arts District on State and Frink Streets. Public art is essential to this vision and the first step was to issue a Call for Artist for two (2) murals to be painted on private property on State Street. The Call for Artist also asked respondents, if interested, to present an overall concept for developing the City's Arts District. Will Bryan Design submitted a proposal that was all encompassing and would bring art to public spaces, by transforming empty walls into energetic and engaging spaces for the community to participate with and enjoy.

Mr. Bryan's proposal for developing the Cayce Arts District, provides for several key elements, at a minimum: branding; painting blighted buildings; creating multiple murals; installing sculptures and public artworks; and constructing gateway signage.

Mr. Bryan will serve as Creative Director for this effort, but will utilize various artists to accomplish the various tasks.

The cost for the proposed branding and design of the City's Arts District is \$77,600 which could be funded with City Hospitality Tax Funds. Staff has applied for two (2) grants that could be used for this project, as well. If the City is awarded the grant monies the Hospitality Tax Fund would be reimbursed. Mr. Bryan's proposed project scope can be completed within 12 months with the Brand Identity design process beginning one week of the proposal being approved.

Recommendation

Staff recommends Council approve the proposal for developing the City's Arts District in the amount of \$77,600, utilizing hospitality tax funding, and authorize the City Manager to execute the related contract with Will Bryan Design.

APPROVED MINUTES
ZONING BOARD OF APPEALS
CAYCE CITY HALL
1800 12TH STREET, CAYCE SC
Monday, JANUARY 28, 2019
6:00 PM

I. CALL TO ORDER

The meeting was called to order by Chair Robert McLeod at 6:00 p.m. Members present were Jason Simpson, Robin DiPietro, and Bob McArver. Russ Vickery was absent excused. Staff present were Carroll Williamson and Monique Ocean.

II. APPROVAL OF MINUTES

Jason Simpson made a motion to approve the minutes from the November 19, 2018, meeting. Robin DiPietro seconded the motion. All were in favor.

III. STATEMENT OF NOTIFICATION

Monique Ocean confirmed that the media and public were notified of the meeting.

IV. PUBLIC HEARING

Variance Request 001-19 and Variance Request 002-19

The property owner is requesting a variance from the **Zoning Ordinance Section 8 Sign Regulations** to:

- Exceed the permitted sign height (7 feet) in the 12th Street Extension Design Overlay District and
- 2. Exceed the allowable number of signs (1) per street frontage.

The property is located at 2350 Foreman Street (TMS 005797-02-006).

a. Opening Statement

Scott Lambert came before the Board as the applicant to explain the variance request. Mr. Lambert explained that he wished for his sign to exceed the height limitations in the 12th Street Overlay District. Mr. Lambert explained that his business site sits 10 – 12 feet below the elevation of 12th Street and it would make it impossible for people travelling along 12th Street to see a sign at the height allowed by the ordinance. Mr. Lambert stated that he does not wish for the sign to exceed size requirements but he just wants it elevated for visibility. Mr. Lambert also spoke about his request to have 2 signs along the street frontage on Forman Street. Mr. Lambert explained that the brewery provides a shared driveway for the parcel next to his and for the city owned pump station. Mr. Lambert stated that because of the shared driveway, he needed two signs to separate the customer entrance from the service entrance.

b. Public Testimony

ZONING BOARD OF APPEALS APPROVED MINUTES JANUARY 28, 2019 PAGE 2 OF 2

There was no one from the public to speak for or against the variance request.

c. Adjourn Public Hearing

With no further discussion, the hearing was adjourned.

V. MOTION

Variance Request 001-19

Ms. DiPietro made a motion to approve the variance request to exceed the height of the sign by 7 feet above the average height of the road barrier adjacent to the sign location. Mr. McArver seconded the motion. All were in favor. Mr. Simpson made a motion to approve the variance request.

Variance Request 002-19

Mr. Simpson made a motion to approve the variance request to allow one additional sign and exceed the permitted number of sign allowed per street frontage. Ms. DiPietro seconded the motion. All were in favor.

VI. NEW BUSINESS

- a. Nomination and election of Officers for 2019 Mr. McArver made a motion to approve Robert McLeod as Chair and Jason Simpson as Vice Chair. Mr. Simpson seconded the motion. All were in favor.
- b. Nomination and election of Monique Ocean as Secretary for 2019 All were in favor to elect Ms. Ocean as secretary.
- Approval of meeting schedule for 2019
 All were in favor of the meeting schedule.

VII. ADJOURNMENT

Ms. DiPietro made a motion to adjourn. Mr. McArver seconded the motion. All were in favor.

A quorum of Council may be present. No discussion or action on the part of Council will be taken.



APPROVED MINUTES PLANNING COMMISSION CAYCE CITY HALL 1800 12TH STREET, CAYCE SC Monday, August 19, 2019 6:00 PM

I. CALL TO ORDER The meeting was called to order by Chair Ed Fuson at 6:00 pm. Members present were Maudra Brown, Robert Power, and Stockton Wells. Chris Jordan, Chris Kueny, and Joe Long were absent excused. Staff present were Carroll Williamson and Monique Ocean.

II. APPROVAL OF MINUTES

Maudra Brown made a motion to approve the minutes of the July 15, 2019 meeting. Robert Power seconded the motion. All were in favor.

III. STATEMENT OF NOTIFICATION

Monique Ocean confirmed that the media and public had been made aware of the meeting.

IV. PUBLIC HEARING – Text Amendment 003-19

A Request by Staff to amend the Zoning Ordinance *Section 7.5 Temporary Uses* to add language concerning Halloween-related events.

a. Opening Statement

Carroll Williamson began by explaining that the City has been getting requests for Halloween related events such as haunted houses. Mr. Williamson explained that the requests have been for a specific time of year so Staff feels it would be best suited to add to the temporary uses section of the Ordinance. Mr. Williamson explained that a permit would be needed for each event and the venue must meet building and fire codes.

b. Public Testimony

There was no one present to speak for or against the Text Amendment 003-19.

c. Adjourn Hearing

With no further discussion, Mr. Fuson adjourned the hearing.

V. MOTION – Text Amendment 003-19

Maudra Brown made a motion to recommend Text Amendment 003-19 to Council for approval. Stockton Wells seconded the motion. All were in favor.

VI. PUBLIC HEARING – Text Amendment 004-19

A Request by Staff to amend the Zoning Ordinance *Section 6.2 Purpose of Districts and Section 6.10 Design Overlay District* to add language concerning a new Arts Design Overlay District. The amended language imposes specific design criteria and encourages live-work spaces within one section of the City.

a. Opening Statement

Carroll Williamson began by stating that the Art Overlay District is meant to encourage an arts community and allow a live work area. Mr. Williamson explained signage in the area would have more flexibility for wall signs. All existing uses will be permitted to continue and underlying zoning will not be affected but there is a list of outright prohibited uses for the area. Mr. Williamson stated there would be no minimum parking requirements.

b. Public Testimony

Public testimony was heard from Mr. Carl Sherwood, Mr. John Bodie, and Mr. Doug Burnaski among other members of the public. The concerns from the public speakers included: The requirement for new businesses to have all parking in the rear or side may be difficult to meet for an existing structure, the fact that no minimum parking is required may cause problems, and the 200 ft. depth of the district is not desirable because it could leave unusable space in larger parcels.

c. Adjourn Hearing

With no further discussion, the hearing was adjourned.

VII. MOTION - Text Amendment 004-19

Stockton Wells made a motion to recommend Text Amendment 004-19 with changes, including new construction must have parking in the rear or side yard and properties that are contiguous and greater than 200 ft. can request to the Planning Director to be moved into the district. Maudra Brown seconded the motion. Ed Fuson, Maudra Brown, and Stockton Wells voted in favor of recommending the text amendment with changes. Robert Power voted against.

VIII. PUBLIC HEARING – Map Amendment 005-19

A request by Staff for a zoning change to create a new Arts Design Overlay District located on the following parcels:

004655-04-015	005767-01-003	005767-02-016	005767-07-001	005769-01-002
004655-04-016	005767-01-005	005767-02-020	005767-07-002	005769-01-006
004655-04-017	005767-02-006	005767-03-001	005767-07-003	005769-08-001
004655-04-018	005767-02-007	005767-03-002	005767-07-004	005769-08-004
004655-04-019	005767-02-008	005767-04-008	005767-07-006	005769-08-013
004655-04-020	005767-02-009	005767-05-002	005767-07-007	005769-09-001
004655-06-001	005767-02-010	005767-05-004	005767-07-015	005769-09-002
004655-06-002	005767-02-011	005767-05-005	005767-07-016	005769-09-003
004655-07-002	005767-02-012	005767-05-009	005767-07-022	005769-09-004
004655-07-003	005767-02-013	005767-06-001	005767-08-001	
004655-08-001	005767-02-014	005767-06-004	005767-09-001	
004655-08-002	005767-02-015	005767-06-009	005769-01-001	
- • • • • • • • • • • • • • • • • • • •				

a. Opening Statement

Carroll Williamson explained that Map Amendment 005-19 is the follow up to the text amendment to create the new arts design overlay district on the zoning map.

b. Public Testimony

No one from the public spoke for or against Map Amendment 005-19.

c. Adjourn Hearing

With no further discussion, the hearing was adjourned.

IX. MOTION - Text Amendment 004-19 and Map Amendment 005-19

Stockton Wells made a motion to recommend Map Amendment 005-19 to Council. Maudra Brown second the motion. Ed Fuson, Stockton Wells, and Maudra Brown voted in favor. Robert Power voted against.

X. **NEW BUSINESS**

Carroll Williamson informed the Planning Commission that a new Comprehensive Plan update session will held at the next meeting.

XI. ADJOURNMENT

Stockton Wells made a motion to adjourn. Maudra Brown seconded the motion. All were in favor.

A quorum of Council may be present. No discussion or action on the part of Council will be taken.

CITY OF CAYCE EVENTS COMMITTEE MEETING MINUTES Council Chambers August 22, 2019

Present: Danny Creamer, Maxine Creamer, Cindy Pederson, Frank Dickerson, Dave Capps, Johnathon Moore, Alexis Moore and Evony Reed

Absent, Excused: Rachel Scurry, Tevin Spruill, Teresa Mitchell

City Representatives Present: Katinia Taylor and James Denny

Review of Minutes – June 13, 2019

Chairperson Danny Creamer called the meeting to order. Mr. Capps made a motion to approve the minutes as written. Mr. Dickerson seconded his motion which was unanimously approved.

Cayce Fall Fest

Mr. Moore stated that the sub-committee did not have any updates at that time but that the main focus for the Cayce Fall Fest should be marketing the new event. Mrs. Taylor informed the committee that posters and flyers were in the process of being printed and digital billboards will be in place in either Charlotte or Greenville.

The committee discussed as a whole placements of all signs and posters. It was suggested to order at least 20 real estate sized signs to place throughout the different neighborhoods and intersections in Cayce and one large sign to be displayed at Granby Gardens Park.

Mrs. Taylor advised the committee that parking for the Cayce Fall Fest will not be at the Old Busbee School due to the school currently being demolished. She said she would contact businesses on Walter Price Street to see if they will allow event parking on their property.

The next Events Committee meeting will be held September 12th in Granby Gardens Park to finalize the event layout.

Guided Nature Tour

The next Guided Nature Tour is scheduled for Saturday, November 2 with Dr. Rudy Mancke at the N. Ave entrance of the Cayce Riverwalk Park. The first tour will start at 10:00am and the second at 1:00pm. The committee will discuss volunteers at a later date.

Other Business

There being no further business, the meeting was adjourned.

Respectfully submitted,

Maxine S. Creamer